

Article

What Affects the Progress and Transformation of New **Residential Building Energy Efficiency Promotion in China: Stakeholders' Perceptions**

Yinan Li^{1,*}, Neng Zhu¹ and Beibei Qin²

- 1 School of Environmental Science and Engineering, Tianjin University, Tianjin 300350, China; nzhu@tju.edu.cn
- 2 School of Built Environment, University of Reading, Reading RG6 6AW, UK; b.qin@pgr.reading.ac.uk
- Correspondence: lyn.tju@live.com; Tel.: +86-1381-124-0163

Received: 16 February 2019; Accepted: 12 March 2019; Published: 16 March 2019



Abstract: Nationwide energy efficiency (EE) promotion of new residential buildings is affected by multiple factors regarding policies, markets, technologies, capacities, and economics. The perceived influences of these factors by stakeholders are crucial to the effectiveness evaluation of current policies and the selection of policy instruments. However, they are normally assumed or taken for granted. The knowledge gap between stakeholders' perceptions and research assumptions may lead to researchers' recognition bias. Correspondingly, this paper aims to identify the significant factors, perceived by frontline stakeholders, influencing nationwide EE promotion of new residential buildings before 2020 and 2030. Factors were collected through literature review and their influence were evaluated via Analytical Hierarchy Process based on the data collected in the questionnaires distributed to 32 institutes. The theory of Nested Policy Design Framework and Policy Environment was used to structure the hierarchy and generate policy implications. Results indicate that (1) policy factors are of the greatest influence before 2020 and market perfection factors will have great influences from 2020 to 2030, indicating the transformation of governance arrangement to "market-based" and "network-based" from the current legal-based system; and (2) factors regarding market needs are of significant influence in both terms, revealing the way the transformation should be accomplished.

Keywords: analytical hierarchy process; energy efficiency promotion; influencing factors; residential buildings; policy design

1. Introduction

China's unprecedented socio-economic development and fast urbanization (see Figure 1) has driven great expansion of residential buildings. As indicated in Figure 1, the urbanization rate has been increasing steadily and rapidly from 36.22% in 2000 to 58.52% in 2017. The total Gross Domestic Production increased by approximately 7 times from 10.02 trillion Yuan in 2000 to 82.71 trillion Yuan in 2017. In accordance with the fast development of Gross Domestic Production (GDP) and urbanization rate, the area of residential buildings also increased dramatically. According to the data from the Nation Bureau of Statistics of China [1], the area of new residential buildings under construction per year has been increasing by 17% per year on average, compared to the year 2000 level, and reached 12.50 billion square meters in 2014 (see Figure 2). The area of new residential buildings under construction and constructed, after 2014, are leveling out around 12.5 billion square meters and 4.2 square meters, respectively. Along with the gradual improvement of living quality, fast increase of building area, and rapid urbanization, the energy consumption of residential buildings has also been increasing dramatically [2,3]. According to the data from Tsinghua University Building Energy



Research Center, as indicated in Figure 3, building energy consumption has increased by more than two times and reached 814 million tons of standard coal in 2014 whilst it was only 300 million tons of standard coal in 2000. During the same period, energy consumed by residential building rose to 301 million tons of standard coal in 2014 from 131 million tons of standard coal in 2000 [4]. In 2015, energy consumed by residential buildings accounted for 19.10% of total energy consumption (see Figure 4) [5]. Excluding urban centralized heating in Northern China, the energy consumption in residential buildings increased to 199 million tons of standard coal in 2015, which is approximately three times the amount in 2001. Currently, the energy consumption of residential buildings is still growing rapidly. The World Bank predicts that the Chinese urbanization rate will be 70% by 2030 [6]. Experience from developed countries shows that the building industry will account for 30% of annual global CO_2 emissions and 40% of total energy consumption in 2030 [7].



Figure 1. Gross Domestic Production (GDP) and urbanization rate from 2000 to 2017.



Figure 2. New residential building area from 2000 to 2017.



Figure 3. Building energy consumption from 2000 to 2014.



Figure 4. Breakout of China building energy consumption by sector in 2015.

The fast increase of energy consumption has not gone unnoticed and translates into the serious risk of China locking itself in with a large and energy-inefficient housing stock [8]. According to the "13th-Five Year Plan of Building Energy Conservation and Green Building Development" (PLAN) published by China's central government in 2016, all new residential buildings should be built to a standard of 72% energy efficiency (EE) level, compared with the 1980s level, and some advanced provinces have increased this requirement to 75%. Additional policy measures to ensure the realization

of the targets have also been proposed by central and local governments, by eliminating the known barriers in the current policy system, stimulating market-based innovations, improving the capacity of stakeholders developing, and constructing high-performance projects (also known as "Capacity Building" in the China case), as well as accelerating the technology evolvement and its penetration into market. Notably, the EE promotion of new residential buildings is a complex socio-technical system [9]. In addition to the homebuyers, the system involves a large variety of stakeholders concerned with policymaking, marketing, and technology, and it could be influenced by factors that include real estate, construction, buildings, and industry [10–12]. The complexity of the system leads to a difficult question to answer—"perceived by frontline stakeholders, what on earth has significant influence on the development of the EE promotion of new residential buildings?", or in other words, whether these policy packages (or factors included in the policy packages) could affect the industry as expected or if the stakeholders within the industry would respond as assumed.

As for identifying the influencing factors, the existing literature have provided deep insights from both technology and management (policy, market, etc.,) perspectives [11–14]. The results in existing literature has provided abundant data for factor identification for the study, but they are discussed independently in different domains. A panoramic view and a systematic analysis of the new residential building energy efficiency promotion should take all the factors in all the domains into consideration. According to a review by De Boeck, research has tended to focus on the following domains: (1) Area of application and design variables, (2) objective and performance measures, (3) type of analysis [15–17], (4) solution methodology, (5) software tools, (6) case study location, and (7) type of building [18]. Regarding factors influencing the EE promotion of new residential buildings, domain 1 provides a classification of technology factors based on which EE promotion measure is applied and in which part the dwelling it is implemented. Five major groups of technology measures application and corresponding design variables are identified, namely the whole building, the construction parts of the envelope (roof, wall, ceiling, and floor), windows and shading, HVAC systems, and finally, appliance and lightning. Domain 2 and 4 stress the popularity of life cycle studies and indicate the importance of factor identification from life-cycle perspective. Chau et al. summarized the similarities and differences of three streams of life-cycle studies and briefly identified major influencing factors in the construction phase, operational phase, and dismantle/renovation phase [19]. More detailed research regarding factor identification in these phases could be classified into two categories: Practice-based and evaluation system-based. The former focuses on factors of significant impacts in reality [20,21], while the latter pays attention to the factors included in various green building evaluation system, such as LEED (Leadership in Energy and Environmental Design), BREEAM (Building Research Establishment Environmental Assessment Method), or DGNB (German Sustainable Building Concil) [22–24]. In addition, economic factors are often discussed with technology factors [25]. For example, Shi et al. [26] proposed a detailed influencing factors list concerning cost optimization, such as technology R&D cost, marketization cost, and labor cost, according to different stages of construction [17].

From the management perspective, in addition to policy review [27] and status quo [28], research tends to focus on policy instrument evaluation and selection [29], barriers analysis [30,31], and theoretical interacting mechanisms among stakeholders from the perspectives of governments and market. The policy instruments, barriers, and interaction mechanism between policy and markets all could be considered as influencing factors which should be collected in this paper, because they would affect the development of the EE promotion. Representatively, Shen et al. [32] identified major policy instrument for building EE promotion and evaluated their influence based on the comparison of practice in seven countries and regions. Li et al. [28] further analyzed the strengths and weaknesses of current policy instruments. As of barriers, Zhang et al. [33] summarized those major ones and argued that deficiency existed in both policy system and market mechanism field. In addition to mandatory policies, economic incentives are also of great influence yet it is still lacking in current policy system [34]. Market failures, mainly caused by information asymmetries and externalities of

the industry, are identified as the major barriers causing the market to systematically underproduce green buildings [35]. Compared with the relatively independent researches, Zou [36] employed the system dynamics methodology to identify and analyze the risks that could affect building EE promotion market. Risks in six subsystems are involved, namely the laws and regulations, standards and specifications, economy, technology, policy, and education.

As for evaluating the significances of the identified factors, six techniques are of great popularity, namely the economic analysis, the sensitivity analysis, the scenario analysis, single objective optimization analysis, multi-objective optimization analysis, and statistical analysis [18]. The economic analysis is always widely used for the evaluation of a certain technology [15], and for analyzing the propagation of high-performance building technologies [16]. The sensitivity analysis or the scenario analysis is always combined with other types of analysis [37]. The selection between single-objective analysis and multi-objective analysis is normally based on the number of the factors to be discussed. The multi-objective optimization analysis helps identify those factors of significant influence in a complicated system and deals with the optimized choices for multiple objectives [38], which is in line with the focus of this paper. Among the approaches to multi-objective optimization, the analytical hierarchy process (AHP) is widely used in the research regarding building EE promotion [12,39]. For example, Roberti et al. utilized the AHP process to determine the portfolio of optimal retrofits of historic building, by quantifying the intrinsically qualitative conservation compatibility of energy retrofits so that it could be analyzed along with the quantitively expressed energy savings and thermal comfort. This is also what the AHP is good at and the reason why it is adopted in this paper—the AHP provides a comprehensive and rational framework for structuring a decision problem, representing and quantifying its elements, relating those elements to overall goals, and evaluating alternative solutions [40]. The specialty of AHP is its flexibility to be integrated with different techniques like linear programming, quality function deployment, fuzzy logic, and so forth. This enables the user to extract benefits from all the combined methods, and hence, achieve the desired goal in a better way [41]. The AHP approach also has the flexibility to combine quantitative and qualitative factors, to handle different groups of actors, to combine the opinions expressed by many experts, and can help in stakeholder analysis [42].

Notably, most current literature draw conclusions based on international experience [43,44], theoretical analysis [36], or the development history [33]. Despite the great contributions of existing literature, the actual perceptions by stakeholders to the influence of these factors remain insufficiently investigated and analyzed. The identification and the understanding of stakeholders' requirements, expectations, and attitudes towards the influence of these factors are fundamental to the successful implementation of strategies to achieve a higher EE level [45,46]. Making decisions without considering stakeholders' voices may lead to confrontation, dispute, disruption, boycott, distrust, and public dissatisfaction [47]. The neglect, or insufficient consideration of stakeholders' perceptions can lead to the cognitive bias of researchers and decisionmakers [48], and further result in an unsatisfied policy implementation outcome [49]. Defined by Von Winterfeldt and Edwards [50], the cognitive bias is a systematic discrepancy between the "correct" answer in a judgmental task, given by a forma normative rule, and the decisionmaker's or expert's actual answer to such a task. When the biased information/perceptions are used for analysis or decision-making, the quality of modeling and resulting analysis will be seriously reduced. Accordingly, the perceived influences of various factors by frontline stakeholders contribute explanations to the gaps between the expected outcome of policy settings ("the 'correct' answer") and the actual responses from stakeholders ("the 'actual' answer"), by presenting the responsive sensitivity of various stakeholders within the industry confronted with different settings of policy systems.

Contributing to the knowledge gap of the disaccord of perceptions of influencing factor between frontline stakeholders and researchers' recognitions and the lack of stakeholders' perceptions to the influence of factors, this paper aims to identify the factors of significant influence to the nationwide EE promotion of new residential buildings based on the lived experience perceived by relevant

frontline stakeholders. Factors are collected through experts' brainstorming and a literature review and are organized into three hierarchic levels according to the PLAN and the "Nested Policy Design Framework" proposed by Howlett [51]. The short-term (until 2020) and long-term (from 2020 to 2030) significances of each factor are evaluated separately because 2020 is the final year of the 13th five-year plan and most international agreements that China has signed are effective till 2030. The corresponding evaluation process is accomplished by institutional questionnaires distributed nationwide to 32 institutes representing different interests. The final quantitative value of significance of each factor is determined by the results of the analytical hierarchy process (AHP). The corresponding results in each hierarchic level give a stereo and panoramic view of factors, perceived by frontline stakeholders, influencing the EE promotion of new residential buildings in China from the perspective of governance arrangement, policy regime logic, and policy instrument selection for a better policy design and the corresponding implementation outcome.

Specifically, this paper contributes to the literature by answering the following three questions:

(1) *What do stakeholders think in general?* This was asked to elicit detailed descriptions of nationwide investigated stakeholders' opinions regarding the identified factors (described in Section 4.1);

(2) What significantly affects the progress of the EE promotion of new residential buildings? This was asked to provide a thorough identification and to prioritize the influencing factors of the EE promotion of new residential buildings in China on the governance arrangement level, policy regime logic level, and policy instrument selection level on the basis of the AHP method (described in Sections 4.2 and 4.3); compared with the existing literature regarding barriers, this question aims to explore factors both of positive and negative effects.

(3) What will significantly affect the transformation of the EE promotion of new residential buildings in the *future*? This was asked to determine the factors of significant influence in both the short term and long term (described in Section 4.4).

The complete structure and corresponding logic flow of this article is illustrated in Figure 5. The remainder of this paper is organized as follows. Section 2 introduces the novelty of this paper and the theoretical foundation for hierarchy construction, which are based on the nested policy design framework and the policy environment theory. Section 3 presents the detailed process of factor collection, classification, and evaluation. Section 4 summarizes the results of the data processing; discusses the weights and ranking of each criteria, sub-criteria, and factors to provide additional details on how the progress of the EE promotion in new residential buildings is affected; discusses the expectations regarding the situation in the long term; and discusses the achievements and limitations of this study. Section 5 presents the conclusions and corresponding policy implications. By investigating frontline stakeholders, this paper depicts a panoramic and authentic picture of how the nationwide EE promotion of new residential buildings is affected and stimulated. The result helps eliminate the perception disaccord between researchers, policymakers, and frontline stakeholders, and further help them better understand the current situation and the transformation of the EE promotion of new residential buildings from stakeholders' perspectives.



Figure 5. Complete structure and logic flow of the article.

2. The Nested Policy Design Framework and the Policy Environment

The nested policy design framework, proposed by Howlett [51], is a level-based policymaking approach concerning governance arrangement, policy regime, and policy instrument selection

(see Figure 6). Governance comprises actors from all segments of society working together to pursue collective goals [52]. The general category of governance arrangements is legal-based, corporatist-based, market-based, or network-based. Governance arrangement can also be understood from the political (actor constellations), polity (policy regime logic/institutional property), and policy (policy instruments) [53] dimensions. Types of governance arrangements result in different preferences of policy regime logic and policy instrument selection (see Table A2).



Figure 6. Three-level nested framework for policy design.

In the case of China, Zhang, Zhou et al. [52] conducted a detailed governance analysis based on the policy cycle theory raised by Howlett and Ramesh [54] and argued that the legal-based governance continues to dominate in China, although there are elements of market-based governance. This conclusion was also supported by the research conducted by He, Zhang et al. [55], who summarized, classified, and evaluated all the policy instruments used to promote energy efficiency of buildings from 1986 to 2015. The application and impact differences among the administrative, information-based policy instrument and the voluntary agreement were also analyzed in this research. A more detailed analysis was conducted by Shen, He et al., based on a comparison of policy data from seven typical countries [32]. By taking policy resources into consideration in the selection process of a policy instrument, the research further explored the feasibility of each instrument based on international experience and common application strategy of the seven countries.

The policy environment theory is used to identify and classify those factors beyond the full control of the government. In addition to the policy settings, four key fields of factors are proposed: technology progress, optimized economics, market perfection, and capacity building. Technology is the foundation of the EE improvement in housings. The evolvement of technology decides the pace of EE improvement of buildings. Factors of optimized economics affect the massive adoption of energy efficient technologies. The level of capabilities of relevant stakeholders decides the feasibility regarding the implementation of techniques and policies. Factors of market perfection focus on the relationship of various stakeholders. The target of market perfection is to help energy-efficient technologies, products, and applications win in market competition. Above all, drivers in these fields focus on improving the EE level of technologies, industries, housing, and the massive application of energy efficient technology from technical and social perspective.

3. Methodology

The whole methodology comprises three major parts: (1) Factor collection and hierarchy construction, (2) investigation and data collection, and (3) evaluation based on the AHP. The boundaries for factor collection are set at first. Factors were collected according to the literature from the technology and policy perspectives. The classification of the factors was based on the policy environment theory. The structure of the hierarchy was built based on the nested policy design framework. Nationwide institutes' questionnaires were conducted to collect judgements on the influence of these factors. The target group comprised institutes from different climate zones and of different interests. Collected data was processed using the AHP approach. Corresponding conclusions and policy implications were provided based on the calculated weights and rankings. The structure of the methodology and its implementation process are detailed in Figure 7.



Figure 7. Implementation process.

The AHP method is fundamentally based on pairwise comparisons to generate conclusions. Two elements being compared at a given time un-repeatedly reduces the conceptual complexity of an analysis greatly. This simplification involves reasonable assumptions proposed by Satty [56] and the others [57,58]. Given a pairwise comparison, the analysis involves three tasks: (1) Developing a comparison matrix at each level of the hierarchy starting from the second level and working down, (2) computing the relative weights for each element of the hierarchy, and (3) estimating the consistency ratio to check the consistency of the judgment [59].

The AHP method is selected for the following reasons. (1) The AHP converts these evaluations to numerical values that can be processed and compared over the entire range of the problem. A numerical weight or priority is derived for each element of the hierarchy, allowing diverse and often incommensurable elements to be compared in a rational and consistent manner. This specialty helps evaluate those factors quantitively and makes inter-category comparison possible among the factors. (2) Pairwise comparisons and judgements are made by experts. Through a proper selection of experts, this method enables various stakeholders to participate in the evaluation of factors, revealing their effectiveness from an empirical perspective. (3) The AHP method requires a hierarchy and classification of all the factors. If designed properly, the priority ranking of criteria, sub-criteria, and factors (i.e., local and global) can reveal the truth regarding respondents' opinions.

3.1. Factor Collection and Hierarchy Construction

As a complex socio-technical industry, the EE promotion of new residential buildings can be affected by a large quantity of factors from multiple aspects, from the macro economy to technology (Section 2). Therefore, the following principles were applied to limit the range of factors before the identification of specific factors: (1) Factors should influence, or be influenced by, the policies issued by the Ministry of Housing and Urban-Rural Development (MOHURD) in China because it oversees promoting the EE of buildings in urban areas, policymaking, implementation, supervision, and conducting major projects concerning the EE promotion of new residential buildings; (2) factors directly relevant to building EE promotion in single and district scales and extended construction value chains are considered; and (3) factors could influence the interactions between the policy system and market mechanism to eliminate market failure. The first principle defines the boundary from a policymaking perspective, the second principle is from a policy-environment perspective, and the third principle stresses the interaction between policy factors and factors from the policy environment. Based on the literature review, 111 factors were identified and are listed in the Table A1.

With the help of the nested policy design framework, a hierarchy of three levels were designed as follows: Level 1—criteria level-preference of governance arrangement. These criteria concentrate on only the fields and sectors (e.g., market and technology)—that is, "what type of governance arrangement does, or will China have, and what type of policy instruments should be preferred?" is analyzed or answered on this level. Level 2—sub-criteria level-preference of policy regime logic. These sub-criteria concentrate on the subsectors of the EE promotion of new residential buildings— that is, they are trying to build up the foundation for answering the question "which type of instruments should be selected or preferred?" on this level. Level 3: factor-preference of operational plans (technical policy instrument selection). These factors provide the empirical statistics and evidence of parameter configuration of policy instruments for implementation and collaboration among instruments. Figure 8 gives a detailed graphical presentation of the structure of the constructed hierarchy and its components. The correspondence between the different hierarchic levels and the nested policy design framework is also illustrated.

Regarding the design of criteria and sub-criteria, the policy settings (C5), as a major contributor to the improvement of EE level, was first identified. All the policies can be categorized into three types in China: macro policy (C51), operable policy, and technical policy (C54). Plans, decisions, and other supporting industry policies are classified into the macro level because they are not operable; however, they are sometimes directly associated with the building EE promotion industry. Operable policies

can be divided into two sub-categories, namely, monetary policies (C52) and administrative policies (institutions) (C53), because of the different types of policy resources they require.



Figure 8. Graphic presentation of the different levels and sub-categories.

In addition to policy settings (C5), the policy environment theory was introduced for criteria and sub-criteria identification. Influencing factors from a policy environment were classified into four categories according to the literature—technology progress (C4), optimized economics (C3), market perfection (C2), and capacity building (C1). The technology progress (C4) focuses on technical factors, which can be classified into six sub-criteria based on the fundamental theory of heating, ventilation, and

air-conditioning from life-cycle perspective (C41 to C46). In addition, factors associated with promoting EE level and efficient use of resources, such as water recycling, distributed energy, and micro-grids were considered because building EE promotion from the district level is a future development trend (C47). The criteria of optimized economics (C3) focuses on costs (C31) and benefits (C32) from technology innovation, the massive application perspective, and the perspective of operation of enterprises in the industry. The criteria of market perfection (C2) attempts to solve the market failure in the EE promotion of new residential buildings industry [40]. The needs (C22) (from all the market stakeholders and governments), as one of the drivers of market development, was a primary consideration. Additionally, according to the requirements of boundary, other factors influencing the interaction between market and policy systems were considered, such as market vitality (C21) (same as market activity) [41], market expectations (C24), and market mechanisms (C23). The capacity building (C1), defined by Robert B. Hawkins from the perspective of political management, is "a concept that encompasses a broad range of activities that are aimed at increase the abilities of citizens and their governments to produce more responsive and efficient public goods. At its core capacity building is concerned with the selection and development of institutional arrangements; both political and administrative" [60]. Seven capacities were identified based on different actors as follows: guidance (C11) and cooperation (C12), referring to central and local government; implementation (C13), referring to planning, designing, and construction enterprises in the industry value chain; supervision (C14), referring to government and third-party institutes; assessment (C15), referring to third-party institutes; operation (C16), referring to institutions involved in the operation stage of buildings like ESCOs(Energy Service Companies) and facility management enterprises; and innovation (C17), referring to all stakeholders in the industry.

3.2. Questionnaire Design and Data Collection Method

In accordance with the principles of questionnaire design [61], the design of the institutional questionnaire are simple, specific, and easy to understand. However, if strictly following the requirements of the AHP methods, the evaluation of 111 factors in three hierarchic levels will lead to 322 times of pairwise comparisons, which is almost impossible for an expert to finish patiently and effectively. Therefore, the Likert 5-Scale was introduced to evaluate the perceived influence of the criteria, sub-criteria, and factors by the investigated institutes independently—1 is for the least importance and 5 is for the greatest importance. To provide additional details on the future situation, the experts in the target group were required to provide their judgements on the short-term and long-term influence of all the factors.

Nationwide institutional questionnaires were used to collect opinions from various frontline stakeholders. Thirty-two institutions from different types of stakeholders, climate zones, and development levels from the national perspective were asked to fill out the questionnaire. The systematic sampling method was used to select these institutes. Additionally, to further improve the stability and amplify the representation of questionnaires, additional requirements were added depending on the property of the institutes when conducting the institutional questionnaires. These requirements are as follows. For government departments, advice from same-level departments, for example, financial departments, industrialization departments, and quality supervision departments, should also be collected when accomplishing the questionnaire. For other institutions, opinions reflected in the questionnaire should be a combination of opinions from high-level decision-makers, frontline workers, and other individuals involved in the industry. The result of institution selection is in Figure 9.

New residential buildings constructed in different climate zones achieve the same level of EE at different rates. Climate differences also lead to different technical strategies and solutions for achieving the same EE level. As indicated in Figure 9, the selected institutes cover all the three major climate zones in China, namely the cold and severe-cold climate zone, the summer-hot&winter-cold climate zone and the summer-hot&winter-warm climate zone. In addition, three national level institutes were also selected to generate more comprehensive opinions.



Figure 9. Results of institution selection.

From the development-level perspective, cities in eastern China started their career of EE promotion in buildings also in the 1980s and until now, policy systems and relevant technologies have been well-developed, meaning that they have more successful experience than the other two districts. As for western China, affected by "the grand western development program", its industry of building EE promotion develops at an unprecedented pace. Cities/provinces like Chongqing, Yunnan have already formed their own effective policy systems and technological systems which can also provide useful experience. And in middle China, owing to a lack of supporting policy package on the overall level and diversity of climate within the province, much effort was spent on exploring how to coordinate different policies to promote the EE of buildings in one province as a whole.

Additionally, the consideration of institutes' properties was mainly based on the role which various institutes play in the industry. For example, the externality of new residential EE promotion makes it hard for developers to profit from developing new residential buildings of high energy performance. However, it is strongly favored by provincial and municipal authorities due to policy influence and potential benefit, such as increasing job opportunities.

3.3. Data Processing

Before applying the AHP method to the collected data, efforts were taken to transform the independent evaluation based on the Likert 5-Scale into a pairwise 9-scale comparison. The process is as follows:

Step 1: Calculate the mean value (mv_{ik}) of each element in each level. Let v_{ijk} represent the element k's value in the level i from No. j institutions, in which $1 \le i \le 3$; $1 \le j \le 32$

$$mv_{ik} = \sum_{i=1}^{32} v_{ijk} / 32.$$
 (1)

Step 2: Calculate the maximum and minimum value of elements in each level of the hierarchy. Let

$$mv_i^{max} = \max\{mv_{ik}, 1 \le i \le 3\}$$

$$\tag{2}$$

$$mv_i^{min} = \min\{mv_{ik}, 1 \le i \le 3\}.$$
 (3)

Step 3: Normalize each element into 9-scale pairwise comparisons. For the random two elements mv_{im} , mv_{in} in the criteria, assuming that $mv_{im} \ge mv_{in}$, then

$$a_{mn} = round \left[1 + \frac{mv_{im} - mv_{in}}{mv_i^{max} - mv_i^{min}} \times 8 \right]$$
(4)

$$a_{nm} = 1/a_{mn}.\tag{5}$$

With the aforementioned adjustment, the judgement matrix A is formed, filled by the elements calculated in Step 3. The following data processing follows the requirement of the AHP method as follows:

Step 4: Determining of the weight vector. The weight factor is the row factor through ensemble average calculation and normalization of the judgement matrix. Assume that the largest eigenvalue of the judgement matrix is λ_m and the corresponding eigenvector is W_m , the following equation will be achieved:

$$AWm = \lambda mWm AW_m = \lambda_m W_m \tag{6}$$

The complete process to calculate the weight factor is as follows: Sub-Step 1: Calculation of the product of all elements (P_i) in each row

$$Pi = j = 1naijP_i = \prod_{j=1}^n a_{ij}.$$
(7)

Sub-Step 2: Calculation of *n* times squaring root of *P* achieved in Sub-Step 1

$$W_i = \sqrt[n]{P_i}.$$
(8)

Sub-Step 3: Calculation of weight factor by normalization of the vector W

$$w_i = W_i \div \sum_{i=1}^n W_i.$$
⁽⁹⁾

Sub-Step 4: Calculation of largest eigenvalue

$$\lambda_m = \frac{1}{n} \times \sum_{i=1}^n \frac{(AW)_i}{w_i}.$$
(10)

Until this point in this study, the value of w_i represents the weight of each factor and the sum of the weights is 1.

Step 5: Consistency check. The purpose of the consistency check is to verify the reliability of the weight ranking. The larger the difference between the greatest eigenvalue and the number of the elements, the larger the inconsistency. The larger inconsistency can also cause large judgement bias. Thus, the consistency index (CI) is introduced to quantitively measure the inconsistency as follows:

$$CI = \frac{\lambda - n}{n - 1} \tag{11}$$

when the CI approaches 0, the judgement matrix has an increasing consistency, and vice versa. To measure the CI quantitively, the random consistency index (RI) is introduced. Based on the research result from Saaty, the distribution of RI is as follows in Table 1:

Table 1. Distribution of RI.

N	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
RI	0	0	0.52	0.89	1.12	1.26	1.36	1.41	1.46	1.49	1.52	1.54	1.56	1.58	1.59

The corresponding consistency ratio (CR) is defined as follows:

$$CR = \frac{CI}{RI}$$
(12)

The judgement matrix is believed to have a satisfying consistency when CR < 0.1. The corresponding normalized eigenvector can be used as the weight vector. Otherwise the judgement matrix shall be re-built and adjustments to the element shall be made.

Because the Likert 5-Scale was introduced to collect the opinions of stakeholders, whereas the AHP method was used for data processing, the results and discussions are correspondingly two-fold. The results based on the Likert 5-Scale reveals the influence of each factor quantitively. The change of average value between the short term and long term indicates the perceived judgements on each factor independently. However, the other results based on the AHP methods provides more information on the weight and ranking of one factor's influence on the rest—locally and globally—according to the methodology. Therefore, both results are considered when analyzing the collected data.

On the one hand, results based on the Likert 5-Scale are analyzed based on the differences in the interests of the stakeholders. The average scoring of each type of stakeholders and variance (the percentage of maximum difference of average value among stakeholders to the overall average value) between actors are calculated. If the variance is less than 20%, the average value could be considered an agreement among various stakeholders. If the variance is equal to or greater than 20%, it is considered that disagreements among stakeholders are sufficiently significant and require extra consideration.

On the other hand, the results based on the AHP are analyzed based on the calculated weight and ranking of each factor in the three levels. The local priority ranking of all the criteria and sub-criteria indicates the stakeholders' judgements regarding the government arrangement and policy regime logic, respectively. The higher ranking of a certain system is equal to the greater effect of the corresponding factors. The weight and ranking changes on the criteria level indicate the changes that will occur to the governance arrangement in China and suggest what type of policy instruments are preferred under the context of governance arrangement transformation. The weight and ranking of each factor indicate its significance to the sub-criteria level from a stakeholder's perspective. The results also provide suggestions regarding the details of how the preferred policy instrument should be configured. The factor with a higher weight is considered more effective in realizing the expected outcome of a policy instrument. Those factors of higher ranking should be given priority and considered.

4. Results and Discussions

4.1. Brief Results of Investigation

The average opinions of judgements from different stakeholders are summarized in Table 2. All the experts agree on the judgements of the influence of policy settings (C5), market optimization (C2), capacity building (C1), and technology progress (C4) in both terms. The overall opinions (M values in Table 2) indicate that policy settings (C5) have the greatest influence in the short term and the second least influence in the long term. The influence of technology progress (C4) and market optimization (C2) will slightly increase, and market optimization (C2) will result in the greatest impacts in the long term. The changes of C2 and C5 between terms indicate strategic change of the development of EE promotion of new residential buildings in China—that is, a more "marketized" policy environment should be formed. The M increase of C4 stresses the necessity of technology—no matter what type of development pattern should occur; technological progress remains the influence of the same level.

				2016–2020					2020–2030		
		Е	G	RI	Μ	v	Ε	G	RI	Μ	v
	Capacity(C1)	4.250	3.889	4.375	4.160	11.7%	4.250	3.889	4.500	4.200	14.6%
Criteria Level	Market(C2)	4.250	4.182	4.500	4.296	7.4%	<u>4.625</u>	<u>4.182</u>	<u>4.500</u>	4.407	<u>10.1%</u>
	Economy(C3)	4.500	3.800	4.250	4.154	<u>16.9%</u>	<u>4.500</u>	3.455	4.250	<u>4.000</u>	<u>26.1%</u>
	Technology(C4)	4.125	4.091	4.250	4.148	3.8%	4.125	4.364	4.250	4.259	5.6%
	Policy(C5)	<u>4.500</u>	<u>4.273</u>	4.375	<u>4.370</u>	<u>5.2%</u>	<u>4.500</u>	4.000	<u>4.111</u>	<u>4.179</u>	12.0%
	Mean	4.325	4.047	4.350	4.226	7.2%	4.400	3.978	4.322	4.209	10.0%
	C11	5.000	3.889	5.000	4.444	25.0%	<u>4.750</u>	3.889	<u>4.600</u>	<u>4.278</u>	<u>20.1%</u>
	C12	4.250	3.111	4.400	3.722	34.6%	4.000	3.111	4.400	3.667	35.2%
	C13	<u>4.500</u>	<u>3.778</u>	$\frac{4.400}{1.100}$	<u>4.111</u>	<u>17.6%</u>	4.000	3.889	4.600	4.111	17.3%
Capacity	C14	4.500	3.778	4.400	<u>4.111</u>	17.6%	4.000	4.222	4.600	4.278	14.0%
1 5	C15	4.250	3.556	4.400	3.944	21.4%	4.250	3.667	5.000	4.167	32.0%
	C16	4.000	3.778	4.600	4.056	20.3%	4.000	3.889	5.000	4.222	26.3%
	C17	$\frac{4.750}{4.464}$	<u>4.111</u>	4.400	<u>4.333</u>	$\frac{14.7\%}{10.7\%}$	<u>4.750</u>	<u>4.222</u>	$\frac{4.400}{4.600}$	<u>4.389</u>	<u>12.0%</u>
	М	4.464	3.714	4.514	4.103	19.5%	4.250	3.841	4.657	4.149	19.6%
	C21	3.800	3.556	4.200	3.789	17.0%	4.400	3.778	4.200	4.053	15.4%
	C22	4.000	4.333	4.600	4.316	13.9%	4.400	4.333	4.800	4.474	10.4%
Market	C23	3.800	3.444	3.800	3.632	9.8%	4.200	3.778	4.400	4.053	15.4%
	C24	4.000	3.556	4.200	3.842	16.8%	4.200	3.875	4.400	4.111	12.8%
	М	3.900	3.722	4.200	3.895	12.3%	4.300	3.941	4.450	4.173	12.2%
	C31	4.000	4.111	4.600	4.222	14.2%	4.500	4.000	5.000	4.389	22.8%
Economy	C32	4.000	4.111	4.600	4.222	14.2%	4.250	4.000	4.800	4.278	18.7%
	М	4.000	4.111	4.600	4.222	14.2%	4.375	4.000	4.900	4.333	20.8%
	C41	<u>4.500</u>	<u>4.333</u>	4.200	<u>4.350</u>	<u>6.9%</u>	3.500	3.889	4.600	3.950	27.8%
	C42	3.833	4.000	4.200	4.000	9.2%	3.667	4.000	4.200	3.950	13.5%
	C43	4.500	4.444	4.800	4.550	7.8%	4.667	4.667	5.000	4.750	7.0%
Technology	C44	3.667	3.667	4.400	3.850	19.0%	3.500	4.111	5.000	4.150	36.1%
recuriology	C45	<u>3.833</u>	3.667	<u>3.800</u>	<u>3.750</u>	4.4%	4.000	4.111	4.800	4.250	18.8%
	C46	3.500	3.444	4.000	3.600	15.4%	4.167	3.889	4.600	4.150	17.1%
	C47	3.667	3.333	3.800	3.550	13.1%	4.000	3.889	4.600	4.100	17.3%
	М	3.929	3.841	4.171	3.950	8.4%	3.929	4.079	4.686	4.186	18.1%
	C51	4.000	4.333	4.400	4.278	9.4%	4.250	4.000	4.800	4.278	18.7%
	C52	4.750	<u>4.333</u>	4.600	<u>4.500</u>	<u>9.3%</u>	<u>4.750</u>	4.000	<u>4.600</u>	<u>4.333</u>	17.3%
Policy	C53	4.500	4.222	4.600	4.389	8.6%	4.500	4.500	4.600	4.529	2.2%
	C54	4.500	4.000	4.200	4.167	12.0%	4.500	4.125	4.500	4.313	8.7%
	М	4.438	4.222	4.450	4.333	5.3%	4.500	4.156	4.625	4.363	16.7%

Table 2. Average opinions of judgements from different stakeholders.

Notes: E-Enterprises; G-Governments; RI-Research Institute; M-Mean; V-Variance.

In addition to the agreements achieved among various stakeholders, there are still attitude preferences caused by the information asymmetry existing in the EE promotion industry. Different stakeholders hold different opinions about the influence of in each criterial and sub-criterial level. Generally, governments tended to be the most conservative stakeholders about the influence of the identified criteria and sub-criteria. Exceptions are marked as italic in Table 2. Research institutes, overall, gave higher evaluation scorings than the responses from enterprises. The corresponding exceptions are marked as underlined.

As to stakeholders' attitudes towards the influence of each criteria, Figure 10 depicts the corresponding results. In the short term, enterprises believe that the influence of economic factors (C3) is greatest and is in line with the influence of policy settings (C5), whereas governments believe that economic factors (C3) is of the least influence. Research institutes believe that factors regarding market (C2) are of the great influence. In the long term, all the stakeholders believe that the factors regarding market (C2) are of the most significant influence. In addition, research institutes also believe that the capacity building (C1) is of the same influence as the factors in market system (C2). Enterprises hold the opinions that factors regarding economy (C3) are as influential as the factors regarding policy settings (C5). Research institutes believe the economy factors (C3) are of greater influence than policy settings (C5), while governments hold the opposite judgements.



Figure 10. Stakeholders' average opinions on each criterion. (**a**) Depicts their opinions regarding the short-term influence of each criteria and (**b**) depicts the opinions regarding the long-term influence.

However, according to the methodology in Section 3.3, agreements on the importance of these criteria are still considered to be reached because the variance of stakeholders' judgements are smaller than 20%. Under this circumstance, it is considered by the author that whichever stakeholders' opinions differ, they are not significant enough to change the agreements among all the stakeholders.

Notably, significant disagreements are still observed among stakeholders on the criteria and sub-criteria levels, in addition to the attitude's preference. These disagreements are marked as bold in Table 2. Experts representing enterprises and research institutes believe economic factors in C2 have substantial impacts, whereas experts from governments do not. The author asserts that this disagreement is the symbol of information asymmetry in the EE promotion of new residential building industry and the proof of the externalities as described by Reference [35]. The judgements provided by enterprises reveal their eagerness for profits due to externalities, whereas governments believe that economic factors could not influence the EE promotion as effectively as policies (see C5) and the perfection of market mechanisms (see C2). In the long term, this disagreement is further enlarged. Governments' judgements on economic factors' influence have further decreased, and the remaining stakeholders keep the same.

Regarding the sub-criteria level, disagreements are also observed and are mostly in the capacity field (see C11–C17). Enterprises and research institutes believe the capacity of guidance (C11), cooperation (C12), assessment (C15), and operation (C16) will significantly influence the improvement in capacity in both terms, whereas governments do not. These disagreements further support the information asymmetry between governments and markets as mentioned in a previous section. In addition, these disagreements stressed the need for more capable and effective governments from the market's perspective because the sub-criteria C11, C12, and C15 mainly refer to governments. This judgement is further proven by the judgements from research institutes.

As for the disagreement identified in the economy system (see C31–C32), the debate concentrates on whether costs should have great significance or super significance. Consensus could therefore be considered achieved. The disagreements in technology aspects (see C41–C47) focus on the efficiency of building materials (C41) and building operation (C44), which are in consensus in the short term. This result is partly because, currently, China is realizing the EE target of a building by increasing the thickness of insulation, which causes many safety problems regarding structural strength of the envelop and inflammability. A small step in improving the efficiency of building materials could lead to a giant leap in the overall energy efficiency promotion. As for operation, technology measures cannot significantly reduce the energy consumption of residential buildings due to the complexities, such as property rights, residential patterns, and various lifestyles.

4.2. Weights and Rankings of Criteria and Sub-Criteria

The weights and rankings of criteria and sub-criteria are summarized in Table 3. The weights and rankings of all the criteria indicate what governance arrangement dominates at the current stage and should occur in the long term, while those of all the sub-criteria prove the governance arrangement transformation and further indicate how should the transformation be realized.

	2016-2020		2020-2030		- Carlo Caritorio		2016-	2020	2020-	2030	Weight				
Criteria	W	R	W	R	– Sub-Criteria		W	R	W	R	Change				
					Guidance	C11	0.358	1	0.188	3	-0.170				
					Cooperation	C12	0.055	6	0.033	7	-0.022				
Capacity					Implementation	C13	0.097	3	0.067	6	-0.030				
Building	0.053	3	0.085	3	Supervision	C14	0.097	3	0.188	3	0.091				
Dununig					Assessment	C15	0.055	6	0.099	5	0.044				
					Operation	C16	0.097	3	0.121	2	0.024				
					Innovation	C17	0.243	2	0.304	1	0.061				
					Market Vitality	C21	0.142	3	0.106	3	-0.036				
Market	0.2(1	2	0 (22	1	Needs	C22	0.627	1	0.598	1	-0.029				
Perfection	0.261	2	0.623	1	Market Mechanism	C23	0.073	4	0.106	3	0.033				
					Market Expectation	C24	0.158	2	0.189	2	0.031				
Optimized	0.052	2	0.045	-	Cost	C31	0.5	1	0.667	1	0.167				
Economics	0.053	3	0.045	5	Benefit	C32	0.5	1	0.333	2	-0.167				
					Building Materials	C41	0.265	2	0.039	6	-0.226				
T					Building Services and Appliance	C42	0.116	3	0.039	6	-0.077				
Technology	0.053	3	0.185	2	Design Phase	C43	0.432	1	0.515	1	0.083				
Evolvement					Operation Phase	C44	0.074	4	0.09	3	0.016				
					Construction Phase	C45	0.053	5	0.148	2	0.095				
					Dismantle/Renovation Phase	C46	0.032	6	0.09	3	0.058				
					District Perspective	C47	0.029	7	0.08	5	0.051				
					Macro Policy	C51	0.16	3	0.239	2	0.079				
Policy	0 59	1	0.061	4	Monetary Policy	C52	0.467	1	0.433	1	-0.034				
Settings	0.58	0.58	0.58	0.58	0.58	8 1	0.061	4	Policy Mechanism Design	C53	0.278	2	0.239	2	-0.039
					Technology Policy	C54	0.095	4	0.088	4	-0.007				

Table 3. Results of the analytical hierarchy process (AHP) on criteria and sub-criteria levels.

Notes: W—Weight; R—Rank.

4.2.1. Weights and Rankings of Criteria

At the criteria level, policy settings (C5) and market optimization (C2) are of the greatest importance in the short term. The summation of the weights of these two criteria accounts for over 80% of the total weight of the five criteria. This result is in accordance with the fact that the combination of "legal-based governance" and "corporatist-based governance" dominates in the field of the EE promotion of new residential buildings at the current stage, as reviewed in Section 2 and discussed in Section 4.1. The ranking of C2 in second place in the short term also indicates that the market for EE promotion of new residential buildings is still not mature. In addition to market failure, many problems must be solved by policy settings based on a mandatory method such as administrative licensing [62–64]. In the long term, the influence of policy settings of market optimization (C2) rises to first. The ranking shift implies that the current developing mode will be substituted by the new mode "guided by government, predominated by market entities" in the long term, and the governance arrangement will shift toward "market governance" and "network governance."

The criteria of capacity building (C1), optimized economics (C3), and technology evolvement (C4) have an identical ranking in the short term, which reveals their equal importance to the EE promotion under current governance arrangement. In the long term, the ranking of C4 rises to second while that of C3 decreases to fifth. The ranking change of C4 between terms shows the importance of technology

factors. Under the background of stable and effective market mechanisms and policy settings, the continuous development and popularization of technology will become a strong support for the industry. As for C3, the author asserts that, the identical rankings in both terms are determined by market stakeholders' continuous eagerness of benefits. This argument is also proved in Table 2—that C3, evaluated by enterprises, is of the same importance in both terms. In addition, the weight increase implies that economy will play a more important role in the long term.

4.2.2. Weights and Rankings of Sub-criteria

The forecasted governance arrangement transformation does not lead to substantial ranking changes of most sub-criteria (except C23 and C51) in C2 and C5, as indicated in Figure 11. It is asserted by the author that the consistency of these criteria between terms are determined by the role that each sub-criterion plays in the EE promotion process. As for the sub-criteria in policy setting (C5), the monetary policy (C52) and policy mechanism design (C53) rank stably front due to their effectiveness in satisfying market stakeholders' eagerness for profit (C31, C32). Technology policy (C54) ranks last since it has no incentives for stakeholders. Regarding market perfection (C2), market needs (C22), as the source power of market development, stably rank first in both terms. Market expectations (C24), depicting stakeholders' willingness participating in the building EE promotion market, ranks second in both terms. Market vitality (C21), as the sub-criterion describing the real activeness of the market, ranks third in both terms with a decreased weight.



Figure 11. Weights and rankings of each sub-criterion.

However, the weight change of each sub-criterion still reveals how the policy setting will change and how the market should be perfected. For example, the weight decrease in policy mechanism design (C53), along with the weight increase in market mechanisms (C23), stress the long-term market perfection through improved policy settings (C52) in the short term. The weight and ranking increases in macro policy (C51) also indicate the importance of market expectations (C24).

The substantial ranking changes in capacity building (C1) criteria depict the importance of various kinds of capacities regarding difference stakeholders. In the forecasted market-based/network-based governance arrangement, governments play a role in eliminating market failures. The opinions and perceptions of market stakeholders are expected to dominate, and guidance from governments is not a necessity. Therefore, the government's capacity of guiding (C11) will not be as important as they are now. The capacity for supervision (C14), as a policy instrument adjusting market activities, remains of great importance in both terms. The corresponding weight increase indicates its increasing importance in eliminating market failures with the change to the new governance arrangement. The capacity of assessment (C15), defined as the capability to evaluate the quality and the amount of building EE promotion, will be of great influence and a prioritized ranking because it is the foundation for the development of "energy efficiency-oriented market mechanisms." The capacity of innovation (C17)

ranks second and further rises to first place in the long term. The author asserts that this ranking is caused by the specialty of China situation—the specialty of overall governance arrangement and the residence pattern lead stakeholders to little advanced experience that can be referred to. The encountered barriers and problems need to be solved innovatively according to specific situation.

The ranking changes of technology evolvement (C4) also indicate the technology strategy change for a higher efficiency in the new residential building sector. In the short term, the EE level in the design phase (also known as energy efficient design) accounts for the biggest weight. This ranking shows the importance of energy efficient design and reveals that, in the current stage, the throttle of EE promotion before construction is the building materials. The ranking of appliances and operation implies that space remains for the reduction of energy consumption by renovations at a low cost or the behavior of residents. In the long term, this priority ranking becomes EE promotion in the design phase > construction phase > operation phase = dismantle/renovation phase > building materials = building services and appliance. The new ranking stresses the importance of construction and shows that the influence of the EE performance of building materials and appliances will significantly decrease.

4.3. Independent Analysis of Factors

The calculation results of all the factors are listed in Table A3. The weight and ranking of each factor indicate their importance and relative priority in a sub-criterion perceived by experts. Instead of simply describing the meanings of weight and ranking of each factor, this section concentrates on the factors with significant influence, the corresponding empirical proofs, and the relations between these factors. Those factors with a higher weight than the average, in either term, in a cluster are identified as significant. The results are summarized in Figures 12–16.



Figure 12. Evaluation results of significant factors in policy settings criteria.





Figure 13. Evaluation results of significant factors in market perfection criteria.



Figure 14. Evaluation results of significant factors in optimized economics criteria.

Figure 15. Evaluation results of significant factors in technology evolvement criteria.



Figure 16. Evaluation results of significant factors in capacity building criteria.

The following discussions will start from the significant factors in the policy system (C5) as it ranks first in the short term on criteria level, followed by market system (C2) as it ranks second in the short term and has the greatest weight in the long term. The factors in economic criteria (C3), as a connection between market and technology, will be discussed afterward. Factors in technology criteria (C4) will be analyzed relatively independently to find the routine for future development. Finally, the factors in capacity building criteria (C1) will be discussed in association with the aforementioned criteria as it involves all the stakeholders in the construction value chain.

Policy Settings (see Figure 12): National mid-/long-term development plan (F511) is of significant importance in both terms since it plays an important role in establishing market expectations (see F241). The significance of industry-supporting policy (F514) is determined by the complexity of the EE promotion industry as a crossover industry among production, construction, and operation. The industry-supporting policies focus the whole construction value chain instead of one key link. Correspondingly, the factor "floor-area ratio reward" (F525) is evaluated as the most effective policy instrument by experts because it is normally awarded to the developers. By increasing the profit of the developers who are in the font stage of the value chain, the value will be further conducted to the downstream industry [65,66]. Additionally, the high housing price makes it easier for developers and downstream stakeholders to attain more profit than other current monetary policies in China.

The factor "local government assessment system" (F532) is evaluated as significant in both terms due to its effectiveness in stimulating local governments' willingness to participate in the EE promotion of new residential buildings. The assessment result will directly influence the human source allocation of local governments and the total amount of subsidy from central governments.

For technology policy (see C54), the revision of codes and standards (F541) is significant because codes and standards draw the EE baseline of design and construction of high-performance buildings, and their renovation speeds determine the pace of technology evolvement of the whole industry. The diversity of methods realizing the requirements of the standards determines that the renovation of atlas is not such a profound influence as the renovation of codes. Additionally, because of the property of compulsion, the renovation of codes, standards, and atlas have greater influence than the recommended list.

Market Perfection (see Figure 13): Regarding the specific factors in the "needs" field, the needs of governments and homebuyers (see F221, F222 in Figure 13) dominate. In the short term, the building EE promotion market is driven by the need of governments to simply fulfill their responsibility, while in the long term, homebuyers will play the most significant role in the process. The ranking change of these two factors is also the proof of the governance arrangement transformation.

The weights and rankings of factors in the other three sub-criteria depict the details regarding how they interact with each other. The stability and continuity of policies (F241) will significantly

increase the market expectations of various stakeholders [67]. After the ignition of market entities' willingness, a good policy environment (see F212) can be a powerful support for stakeholders to develop continuously in this field. Governments shall also take actions to establish effective business modes (F233) and financing channels (F232) for the enterprises in the short term. In accordance with the governance arrangement transformation discussed in Section 4.1, the influence of factors introducing market investment (an identical ranking change of F213 and F233) will also increase in the long term. The availability of financing products, services, and channels (F213) will become of great significance and affect the stakeholders' active participation in the market. Additionally, the buildup of an effective market mechanism should start with taking advantage of energy efficiency trading (F234) because successful pilot projects have already been observed in Shenzhen.

Optimized Economics (see Figure 14): In the cost field, the cost of technology R&D (F311) plays the most important role in the short term. In accordance with the conclusions and speculations on the analysis of the market system and capacity system (F173), this ranking indicates that the promotion of technology shall be conducted, and an upgrade of the technology system is a necessity. In the long term, the level of marketization of technology and products (F314) becomes primary in the cost sub-criteria, which is logically in line with the theory of the technology life-cycle [68]. The payback period (F322) in the benefit part remains significant in both terms. On the one hand, the significance of F322 is in accordance with the ranking of the cost because technology R&D and marketization are tightly associated with the calculation of the payback period; on the other hand, in China's residential building sector, despite the amount of EE promotion, a relatively low energy price [69] leads to a longer payback period for enterprises, concerning facility management or energy management.

Technology Evolvement (see Figure 15): The design phase, as an early stage of the building's life-cycle, plays a critical role in determining the EE level of the architecture in both terms. Factors regarding thermal performance of windows (F413, F432, F433) rank front, followed by the direction the building faces (F437) and natural ventilation (F436) in both terms. In accordance with the factors' evaluation in the building material field (F411), the strategy of energy efficient design can be described as follows. "For the non-transparent part, the EE improvement should focus on the thermal performance of insulation, instead of simply increasing the thickness. For the transparent part of a building (see F412), its design should comprehensively take the layout of the buildings, certain room and the thermal performance of materials into consideration."

For the other technology aspects of newly constructed buildings, the concept of "clean and renewable energy/resources utilization" (F444, F449, F473, F475, F447) dominates in the short term, and the concept of "energy/resources recycling" (F445, F440, F462) dominates in the long term. This ranking of factors in the construction (see C45), operation (see C44), and renovation (see C46) fields means that, despite the energy and resource shortage in China, sufficient no-fossil energy/high-grade energy remains for the consumption of residential buildings in the short term. The comprehensive use of various types of energy and resources will increase the EE level because the evaluation is normally fossil-energy/non-renewable resources-based. However, the exacerbation of energy and resources will finally affect the energy consumption of housing in the long term. The strategy will then shift to "recycling" rather than "comprehensive utilization."

Capacity Building (see Figure 16): Governments always play an important role in both terms. In the short term, the governments' ability to establish the implementation standards (F113) ranks first. The implementation standard, under this context, refers to instructing market entities regarding how to realize the EE target by setting up a conditional incentive system and technological specifications. The more detailed the implementation standard, the more extensively the government dominates. In the long term, a government's ability to guide the energy-saving consciousness (F111) ranks first. Along with the decrease in the influence of the cooperation ability between governments (F122), the ability to guide cooperation market entities (F121) is of more significance in the long term. The shift of this ranking also shows that "market governmence" and "network governmence" will gradually form. In addition to the enforcement of the energy-saving consciousness of stakeholders (see F111), the

building up of a stable cooperation relationship among market entities (see F112, F121, F122) is also important.

Regarding the stakeholders in the construction value chain, the ability of the design construction (F131, F132) of the same importance are observed in both terms. The weight of the factor "capacity of facility management" (F161) ranks first in the management field and increases to approximately 15%. Due to the application of various energy-saving technologies, the operation quality of these pieces of equipment directly affect the energy consumption of a certain building.

With respect to the relevant third-party institutes, Figure 16 presents the details of the capacity building of supervision and assessment. The construction quality (F141), as a basic and mandatory requirement of a building, remains of significant importance in both terms. The capacity of credit supervision (F142) is also evaluated as significant since market stakeholders will dominate the EE promotion process in the long term. The capacity of assessing energy saving (F152) is evaluated as significant in both terms since the quantified evaluation of energy saving is the foundation for the design of incentives. Additionally, that the innovation ability (C17) ranks second indicates its importance in promoting the EE level of buildings. In accordance with the ranking change of a technology system on a criteria level, the ability of technology innovation (F173) ranks first. Technology, as the foundation of the EE improvement, directly determined the highest level of the promotion.

4.4. Comprehensive Analysis of All the Factors

Other sections have discussed the weights, rankings, and corresponding implications of each factor in a separate sub-criteria system. However, chances remain that one factor in a certain system of minor influence may have a great global influence. Additionally, the continuity of one factor having a significant impact in both terms will help the policymakers improve their understanding of the building EE promotion process in China. Thus, all the factors are analyzed in this section. Weight and the corresponding global ranking of each factor is calculated, neglecting its belonging field. Figure 17 is a plot of all the factors based on their rankings in both terms.



Figure 17. Distribution of all the factors.

As shown in Figure 17, factors in the "market perfection system" (C2) and "policy settings system" (C5) tend to cluster (as indicated in the blue and gray circles), whereas the distribution of the factors in the remainder of the systems is decentralized. Factors in the policy settings criteria (C5) have a higher ranking in the short term, and the factors regarding market perfection (C2) have a higher ranking in the long term. Factors in the optimized economics system (C3) and capacity building system (C1) are generally located within a range of 40–100 in the short term and 20–100 in the long term, as indicated by the color orange in Figure 17. Most of the factors in the technology system (C4) have even lower

rankings between 60 and 120 in the short term or between 80 and 120 in the long term (blue area in Figure 17).

As shown in Figure 18, few factors are of great weight, and most factors have weights less than 0.020. To focus on the factors of relatively great influence and to cover at least 80% of the weight of the total, the accumulative weight based on the top-down ranking was calculated. Correspondingly, the value at which the accumulative weight exceeds 0.8 for the first time was taken as the baseline for factor selection. Now, 53 factors (i.e., 16 from short term specifically, 25 from long term specifically, and 12 overlaps in both terms) remain on the list. The calculation results of effective factors are listed in Figure 19.



Figure 18. Top-down ranking of factors in the short and long term; (**a**) presents the ranking in the short term and (**b**) presents the ranking in the long term.

Most factors specific in the short term belong to the policy setting system (C1). The corresponding range covers all the fields of the policy settings. The only exception in this category, the factor of setting up an implementation standard (F113), is also "government relevant." The distribution of these factors indicates the importance of government and policies in the short term.

As to factors in the long term, the span covers the systems of capacity building, optimized economics, market preference, and technology. The diversification of these factors implies that more stakeholders will take part in the promotion process under the framework of market governance or network-based governance. Instead of the diminished influence of factors regarding policy settings, factors concerning market optimization (C2) are of great importance in the long term. Additionally, other long-term factors, except those regarding the technology field (C4), are also market-relevant.

Moreover, a total of 12 factors overlap in both terms, covering the fields of policy setting (C1), optimized economic system (C3), and market system (C2). As a bridge between the current stage and the future, these factors, significant in both terms, will play a continuous role in the next 15 years in the process of building EE promotion. The high significances of these factors also provide a solid foundation and continuity for the transformation and the development of the Chinese market. Furthermore, most of the market factors belong to the "needs" sector. As one of the drivers of building EE promotion, this need for EE promotion of new residential buildings will continue to be in an increasing and transforming process, and the EE level in the residential sector must be promoted for, at least, the next 15 years. The increase in the number of factors that remain on the list in market and economics and the decrease in policy systems prove that China is shifting its development pattern into a "market-dominated" mode. The increase in the number of factors of capacity building systems proves that this pattern shift, and future building EE promotion development, should be led and cultivated by the government, which is quoted in Section 5 as "guided by governments."

	4 0		
	4 F4	Construction Waster Recycling	⁹² ²¹ Factors in the Technology
	151 151	Industrialized Production of Green Building Material	⁸⁹ ²¹ Factors in the rectinology
	33	Thickness of Insulation Material	81 20
	F4	Green Construction Technology	
	42 42	Credit Supervision of Market Entities Behavior	
	F2 31	Trading Mode	
	F4 38	Natural Illumination	68 22
	F4 35	Limit of Building Shape Coefficient	6 7 ³ 2
SIC	F1 41	Supervision of Construction Quality	F $_{61}^{29}$ Factors in the Capacity
acto	F1 61	Capacity of Facility Management	F 60 ³⁴ Building System
СF	F1 11	Guidance of Energy–Saving Conscious	4 9 10
scifi	F4 37	Facing Direction of the Building	• 48 19
Spe	F4 36	Natural Ventilation design	Factors in the Technology
rm	F4 33	Window–Wall Ratio	46 Evolvement System
-Te	F4 32	Types of Windows	45
Suc	F1 72	Innovation Capacity of Business Mode	4 4 ²⁷
Ľ	F2 13	Availability of Financial Products and Services	-43 -15 Factors in the Optimized
	F3 13	Level of Marketization of Technology and Product	42 Fconomic System
	F2 14	Market–based Credit System	• 37 ²⁶
	F1 73	Innovation Capacity of Technologies and Products	3 6 ¹⁷
	F2 43	Evaluation of Market Risks	35 18
	F2 33	Business Mode	34 16
	F2 32	Financing Channel	⁷ Factors in the Market
	F2 34	Energy Efficiency Trading Mechanism	Perfection System
	F2 11	Energy–Saving Consciousness of Relevant Enterprises	3125
	F2 42	Market Potential of Renovation	27 9
ns	F2 25	Energy–Saving Needs from Energy Suppliers	25 11
err	F2 26	Energy–Saving Needs from Energy Users	24 6
th J	F3 22	Length of Payback Period	$\frac{37}{22}$
Bo.	F2 12	Policy Environment Concerning Energy Saving	Pactors of great significance
for	F2 24	Energy–Saving Needs of Occupants	¹⁰ ₅ in both terms
tors	F2 41	Stability and Continuity of Policies	- 3
Fact	F2 23	Enterprises' Needs of Making Profit	- 4
on	F2 21	Energy–Saving Needs of Homebuyers	8 1
นน	F2 22	Governmental Needs of Conducting Work	- 2
Col	F5 21	Financial Subsidy	- 28 2
	F5 25	Floor Area Ratio Reward	23 1
	F1 13	Setting up Implementation Standard	- 58
	F5 36	Copyrights Protection	
	F5 12	Building Energy Efficiency Promotion and Green Building	23
	F5 13	Decisions of Central Government Concerning Building.	
OIS	F5 35	Top Runner Institution	-92^{-1}_{20}
act	F5 42	Revision of Construction Method and Engineering Atlas	87 19
fic F	F5 34	Technical Encouragement and Reward Policies	- 91 16
ecit	F5 11	National Mid/Long-term Development Plan	57 15
ı Sp	F5 31	Industrialization Policy	Factors in the Policy
ern	F5 24	Reduction of Cost of Infrastructure	Settings System
t-T	F5 23	Tendency Loan	
hor	F5	Revision of Codes and Standards	<u>60</u> 10
S	F5	Market Access System	73 7
	F5 14	Industry Supporting Policies	<u>- 42</u> 6
	F5	Assessment and Evaluation system concerning governments	51 5
	F5	Tax Reduction	<u>52</u> 3
	pada C.A.		
	Lo	ng-term Influence of Factors	0 0.02 0.04 0.06 0.08 0.1 0.12 0.14 0.16
		Notes: The label presents	s the corresponding global ranking of the factor
	She	ort-term influence of Factors	1 00 0 0 0

Figure 19. Factor selection results of significant influence.

4.5. Future Study and Practical Implications of the Study

4.5.1. Limitations and Future Study

By investigating frontline stakeholders involved in the industry of EE promotion of new residential buildings, this paper collected and reported stakeholders' perceptions of the factors influencing the EE promotion of new residential buildings in the case of China. By adopting the "nested policy design framework" and the "policy environment" theory, this paper constructed a structure for classifying and organizing the collected influencing factors. The perceived influence of each factor was evaluated by frontline institutes via Likert 5-Scale, and the corresponding weight and ranking was further determined through the AHP process. The investigation and data processing results were validated based on the history of the development, and the corresponding policy implications are further derived.

However, due to the complexity of the industry, the data collected from 32 institutes are still not sufficient for a detailed analysis from each stakeholders' perspective. This paper listed eight categories of stakeholders in Figure 9 but classified them into three categories—government departments, enterprises, and research institutes in the following analysis. The corresponding implication also limit its range within the interactions policy settings and the response from the frontlines. It is considered by the author that too few institutes' voices cannot represent the overall attitudes of the same kind of stakeholders. More and carefully selected institutions involved in the study will provide more useful and authentic information about the perceived influence of factors by different stakeholders. Correspondingly, the question "what stakeholder thinks in detail rather than in general?" can be answered, by taking relevant theories, such as the gaming theory, into consideration. The results can thus help researchers and decisionmakers better understand the difference of response from different stakeholders the same policy system.

Despite the aforementioned limitations, the current results can still be considered as effective due to current governance arrangement in the investigated institutes in China. Once the questionnaire is delivered to a certain institute, managers (college deputies and heads of other government departments and public institutions) will break it down and send them to different departments. Experts in each department will first fill in the questionnaire based on their perceived experiences and their majors. A follow-up workshop, where all the responses from experts and corresponding reasons are stated and discussed, will be conducted to find the most agreed-upon EE level. When approved by the manager of the institute, the most agreed-upon EE level will be the formal response. Therefore, a formal response could be seen as the result of negotiations among experts of different research domains and professional ranks.

4.5.2. Practical Policy Implications

In the following 15 years, the current governance regarding new residential building EE promotion in China will shift to "market-based" and "network-based" from a "legal-based" governance. The industry will correspondingly be guided by governments and dominated by market stakeholders. This transformation also indicates that the building EE promotion process will be more sensitive to diversified factors regarding marketization, namely, market perfection, technology, economy, and capacity building.

With respect to the preference of policy instrument selection, marketization is one of the future trends under the changed framework of governance arrangement. Despite the decreasing influence of a policy system, the monetary policy and policy mechanism remain of great influence because they are effective policy instruments for the ignition of market development and building up market expectation, as perceived by stakeholders within the industry. "Needs", as a basic driver for market development, are the most significant influence on the promotion. Factors of market expectation decide the willingness to participate in the promotion process. Due to the increasing awareness of the externality, market entities are less sensitive to benefit, but remain sensitive to the cost of various types. Regarding the ability to fulfill the corresponding duties of respective stakeholders, the capacity

of governments is of great importance in the short term, and the innovation ability of stakeholders becomes more important because it is one of the core competences of enterprises in the market.

The weight of each factor further indicates the importance of various types of policy instrument selection and specifications. In the short term, governments play a critical role in guiding and cultivating the development of the EE promotion of new residential buildings. With abundant policy resources [70] such as legislation and tax-collecting rights, the government can compensate market failure and adjust industry development along an approach toward higher efficiency under the situation that policy instruments can be selected and designed into a policy package correctly. In addition, evaluation results of factors in the market system indicate that policies should first encourage more enterprises to participate in the EE promotion process by igniting the home buyers' need and satisfying enterprises needs for profits on overall level. When there are more entities participating in the market, it is important to build up a perfect mechanism to further promote the development of the market. In the economy part, factors' weights concerning cost became larger than that of benefits. Current policy should solve the question of the long payback period for incremental cost caused by EE promotion. Regarding the technology system, the strategy of construction can be described as follows: "For the non-transparent part, the EE improvement should focus on the thermal performance of insulation, instead of simply increasing the thickness.; for the transparent part of a building, its design should comprehensively take the layout of the buildings and certain room and the thermal performance of materials into consideration." Additionally, the results of technology evolvement factors imply that the concept of life cycle is becoming mainstream and all the components of building EE promotion tend toward the same importance. Correspondingly, greater attention should be paid to a regulated design, construction, operation, and life-cycle energy saving of buildings (single and district). With respect to capability building system, factors of innovation capacity and supervision capacity concerning regulating and exciting the market show great significance in the long term, implying that building EE promotion should be market-based and market-dominated.

5. Conclusions

Contributing to the gap of the disaccord between stakeholders' perceptions and research assumptions, this paper identified the factors of significant influence to the nationwide EE promotion of new residential buildings before 2020 and 2030 based on the lived experiences of the first-front workers and institutes. Factors were collected through literature review. Questionnaires distributed to over 30 institutes were used to evaluate the influences of all the factors quantitatively and the result was generated from the AHP process. The nested policy design framework and policy environment theory was used to structure the hierarchy and generate policy implications. The corresponding conclusions summarized by answering the following three questions. The stakeholder's attitudes towards various factors regarding EE promotion of new residential buildings presented in this paper helps policymakers, decisionmakers, and researchers better understand the current situation and the future development in China. More internationally, the correspondence between governance arrangement, policy regime logic, and preference of policy instrument selection analyzed in this article helps researchers to understand how the importance of factors will change under different context. The questions are as follows:

What do stakeholders think in general?

Due to the information asymmetry and the different interests that various stakeholders stand for, the perception difference lies not only the general level, but also specifically on the judgements to criteria, sub-criteria, and factors.

Generally, government departments tend to be conservative in evaluating factor importance. The only exception found, on the criteria level, is the importance of technology evolvement in the long term. This stresses governments' attention on the technology evolvement. According to their perceptions, technology evolvement is the foundation of, and decides the pace of, the EE promotion of new residential buildings. In addition, there was no consistent relationship be found between the judgements from enterprises and those from research institutes. On the criteria level, enterprises believe the factors regarding policy settings (C5) and the economy (C3) are of more significant importance, in both the short term and long term, than the research institutes expect. Regarding the judgements on the market criterion (C2), research institutes hold more ambitious opinions about the short-term importance of C2 while enterprises are more ambitious in the long term. Regarding factors on the sub-criteria level, the responses from enterprises are overall more conservative than those from research institutes, although there are six and four exceptions in the short and long term, respectively.

In addition to the attitude preference, significant disagreements can also be observed. These disagreements are further enlarged in evaluating the importance of all the criteria and sub-criteria in the long term. It is considered by the author that these disagreements are caused, not only by the information asymmetry and the externality of the industry, but also by the interests that various stakeholders stand for and the corresponding expectations. The major two significant disagreements are relative to the influence of economic factors and governments' capacity. Enterprises are calling for more profits and they believe governments can effectively affect the development of EE promotion by issuing policies. However, governments believe that the economic factors should have the least influence. Besides, both research institutes and enterprises expect that current government. However, government departments do not believe their capacity are not of such significant influence, as expected.

Despite the attitude preference and the significant disagreements, the agreement on the importance of most criteria, sub-criteria, and factors are still reached. The results of consensus are as follows:

• What significantly affects the progress of the EE promotion of new residential buildings?

Indicated by the weights and rankings of policy settings (C2) and market optimization (C5) in the short term, the "legal-based governance" and the "corporatist-based governance" dominate in the field of EE promotion of new residential buildings. Correspondingly, factors regarding policy settings (C5) have the greatest influence in the short term. The monetary policy (C52) ranks front due to its effectiveness in satisfying market stakeholders' eagerness for profit (C31, C32). The policy mechanism design (C53) has the second greatest importance in regulating market activities and setting up mechanisms for sustainable developments. Technology policy (C54) ranks last since it has no incentives for stakeholders. As for specific factors, those of significant short-term influence are listed in Figure 19 (factors with blue and green cover). The floor-area ratio reward (F525) is of the greatest significant as it stimulates the developers—the stakeholder locating front of the value chain.

It is argued by the author that governments play the most important role in promoting the EE of new residential buildings. With abundant policy resources, they can effectively affect the development of the industry by issuing mandatory policies, setting up implementation standards, and stimulating the activeness of market stakeholders through incentives such as financial subsidies. In addition, it shall be noted that, for local governments, they are stimulated by the requirements from central governments.

• What will significantly affect the transformation of the EE promotions of new residential buildings in the future?

Indicated by the weight and ranking changes of all the five criteria, the current "legal-based" and "corporatist-based" governance arrangements in China in the EE promotion field of new residential buildings will transform into "market-based" and "network-based" governance arrangements. However, this transformation cannot be accomplished suddenly at a specific time. How the transformation should be accomplished or affected, perceived by frontline stakeholder, are illustrated in Figure 19 (factors with green cover).

30 of 41

Generally, the transformation process is affected both by monetary factors and market-related factors. At the beginning, monetary policies (C52) are still of great importance in stimulating the market, as they can directly reduce the incremental cost and increase the benefits of enterprises. However, stakeholders involved in the market activities are calling for a more "stable" market environment. This is also the reason that governments' needs of conducting work (F222), homebuyers' need (F221), and the stability and continuity of policies (F241) are of extreme significant importance in the long term.

Along with the transformation of governance arrangements, factors in more criteria can have significant influence on the future development of EE promotion. Market mechanisms (C23), referring to well-organized competition among stakeholders, are of the greatest importance. The level of marketization of energy efficient building materials and products (F313) determines stakeholder's access to the technology solutions with limited incremental cost. The capacity building of governments (F111, F114, F116) decides how effectively the governments can regulate the market and eliminate market failures. The technology evolvements (F431, F432, F433, F436, F437, F451, F453, F462) indicates how the EE level of technology should be further promoted.

By introducing the nested policy design framework and the policy environment theory, this paper shed further light on the correspondence between governance arrangement, policy regime logic, and the preference of policy instrument selection. The "legal-based" and "corporatist-based" governance arrangement is in accordance with the "government-dominated" development of EE promotion of new residential buildings. Correspondingly, policies issued by governments are of significant importance in stimulating the market. Taking enterprises' eagerness for benefit into consideration, the preferred policy instruments are those monetary ones such as financial subsidy or the floor-area ratio reward. On the contrary, the "market-based" and "network-based" governance involves more stakeholders as the "decisionmakers" in promoting EE of new residential buildings. The development is correspondingly "market-oriented stakeholders-dominated" and governments play a role as a pioneer, a supervisor, and a regulator of the market. Under this circumstance, the EE promotion of new residential buildings can be affected by factors from more aspects than the development mode under "legal-based "and "corporatist-based" governance.

Author Contributions: Conceptualization, Y.L. and N.Z.; data curation, Y.L. and B.Q.; formal analysis, Y.L., N.Z., and B.Q.; funding acquisition, N.Z.; investigation, Y.L. and B.Q.; methodology, Y.L. and N.Z.; project administration, N.Z.; resources, Y.L. and B.Q.; supervision, N.Z.; validation, Y.L. and B.Q.; visualization, B.Q.; writing—original draft, Y.L. and B.Q.; writing—review and editing, N.Z.

Funding: This research was funded by Energy Foundation, grant number G-1402-19981.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Criteria	Sub-Criteria	Factors	No.
		Insulation Performance of Non-Transparent Envelope	F411
		Insulation Performance of Transparent Envelope	F412
	Building Materials Efficiency-C41	Airtightness of Windows	F413
		New type of wall structure	F414
		Solar Shading Devices	F415
		Appliances of higher energy efficiency	F421
	Building Services and Appliance Energy	Choice of Different types of Energy	F422
	Efficiency-C42	Optimized Parameter Settings of Devices	F423
	Efficiency C12	High Efficiency Component of HVAC system	F424
		High Efficiency Lighting system	F425
		Thickness of Insulation Material	F431
		Types of Windows	F432
		Window-Wall Ratio	F433
	Design Phase-C43	Design of Solar Shading System	F434
	Design Fildse Cito	Limit of Building Shape Coefficient	F435
Promotion by Technology Progress-C4		Natural Ventilation design	F436
		Facing Direction of the Building	F437
		Natural Illumination	F438
		Human Behavior Sensors Installations	F441
		Indoor Temperature/Humidity Control	F442
		Indoor Heating/Cooling Energy Consumption Supervision	F443
		Renewable Energy Application on Buildings	F444
	Operation Phase-C44	Energy Recycling	F445
	operation raise err	Distributed Energy Utilization	F446
		New Heating/Cooling sources Utilization	F447
		Building Energy Consumption Supervision	F448
		Comprehensive Utilization of Resources	F449
		Water Saving and Recycling	F440
		Industrialized Production of Green Building Material	F451
	Construction Phase-C45	Construction Process Requirement	F452
		Green Construction Technology	F453

Table A1. Hierarchy and details of assessing factors' weight.

Criteria	Sub-Criteria	Factors	No.
		Construction Waste Reduction	F461
	Dismantle/Renovation Phase-C46	Construction Waster Recycling	F462
		R&D of Construction Waster Utilization	F463
		Reclaimed Water Utilization	F471
		Rian Water Collection and Utilization	F472
	From District Perspective-C47	Underground Space Utilization	F473
	From District refspective C47	Heat Island Effect Reduction	F474
		District Energy Utilization Plan	F475
		Distributed Energy Utilization	F476
		Technology R&D Cost	F311
		Labor Cost	F312
		Level of Marketization of Technology and Product	F313
		Level of Massive Development of Technology and Product	F314
	Cost Optimization-C31	Financing Cost	F315
	cost optimization cor	Promotion of Comprehensive Design Capacity Requirement	F316
Promotion by Optimized Economics-C3		Promotion of Construction Requirement	F317
Fontouon by optimized Economics Co		Technology Application Cost	F318
		Technology Designing Cost	F319
_		Technology Maintenance Cost	F310
		Reduction of Energy Consumption	F321
	Expected Benefits-C32	Length of Payback Period	F322
	Expected benefits C32	Promotion of Building's Value	F323
		Energy-Saving Consciousness of Occupants	F324
		Energy-Saving Consciousness of Relevant Enterprises	F211
		Policy Environment Concerning Energy Saving	F212
	Market Vitality-C21	Availability of Financial Products and Services	F213
		Market-based Credit System	F214
		Third-party institutions Development Concerning	F215
Promotion by Market Perfection-C2		Energy Services	
		Energy-Saving Needs of Homebuyers	F221
		Governmental Needs of Conducting Work	F222
	Needs-C??	Enterprises' Needs of Making Profit	F223
	110000-022	Energy-Saving Needs of Occupants	F224
		Energy-Saving Needs from Energy Suppliers	F225
		Energy-Saving Needs from Energy Users	F226

Table A1. Cont.

Criteria	Sub-Criteria	Factors	No.
		Trading Mode	F231
	Market Machanisma C22	Financing Channel	F232
	Market Mechanisms-C25	Business Mode	F233
		Energy Efficiency Trading Mechanism	F234
		Stability and Continuity of Policies	F241
	Market Expectations-C24	Market Potential of Renovation	F242
		Evaluation of Market Risks	F243
		Guidance of Energy-Saving Conscious	F111
	Cruidanas C11	Cultivation of Market Development	F112
	Guidance-C11	Setting up Implementation Standard	F113
		Guidance of Value Chain Perfection	F114
_	Cooperation C12	Cooperation among market entities adjusted by governments	F121
	Cooperation-C12	Cooperation between Governmental Departments	F122
_	Implementation C12	Capacity of Planning and Design	F131
	Implementation-C15	Capacity of Construction	F132
Promotion by Capacity Building-C1		Supervision of Construction Quality	F141
	Supervision-C14	Credit Supervision of Market Entities Behavior	F142
		Behavior Criterion for Market Entities	F143
-		Capacity of Assessing Construction Quality	F151
	Assessment-C15	Capacity of Assessing Energy Saving	F152
		Capacity of Energy Efficiency MRV	F153
_		Capacity of Facility Management	F161
	Operation-C16	Technical Capacity of ESCOs	F162
	-	Maintenance Capacity of Energy Supply Enterprises	F163
-		Innovation of Cooperation Mode promoted by governments	F171
	Innovation-C17	Innovation Capacity of Business Mode	F172
		Innovation Capacity of Technologies and Products	F173

Table A1. Cont.

Criteria	Sub-Criteria	Factors	No.
		National Mid/Long-term Development Plan	F511
	Macro Policy-C51	Building Energy Efficiency Promotion and Green Building Development Plan	F512
		Decisions of Central Government Concerning Building Energy Efficiency Promotion	F513
		Industry Supporting Policies	F514
		Developing trend of Building Energy Efficiency on Domestic/International Level	F515
Efficiency Promotion Caused by Policy		Financial Subsidy	F521
Settings-C5		Tax Reduction	F522
, and the second s	Monetary Policy-C52	Tendency Loan	F523
		Reduction of Cost of Infrastructure	F524
		Floor Area Ratio Reward	F525
_		Industrialization Policy	F531
		Assessment and Evaluation system concerning governments	F532
	Policy Machanism Docign C53	Market Access System	F533
	Toncy Wechanishi Design-C55	Technical Encouragement and Reward Policies	F534
		Top Runner Institution	F535
		Copyrights Protection	F536
_		Revision of Codes and Standards	F541
	Technology Policy-C54	Revision of Construction Method and Engineering Atlas	F542
		List of Recommended Products and Technologies	F543

Table A1. Cont.

 Table A2. Modes of governances and its implementation preference.

Mode of Governance	Overall Governance Aim	Implementation Preference
Legal governance	Legitimacy and compliance through the promotion of law and order in social relationships	Legal system: legislation, law, and rules and regulations
Corporatist governance	Controlled and balanced rates of socioeconomic development through the management of major organized social actors	State system: plans and macro-level bargaining
Market governance	Resource/cost efficiency and control through the promotion of small- and medium-sized enterprises and competition	Market system: auctions, contracts, subsidies, and tax incentives and penalties
Network governance	Co-operation of dissent and self-organization of social actors through the promotion of inter-actor organizational activity	Network system: collaboration and voluntary associational activity and service delivery

	No. Factors —		Local	Weight	Ranking		
Sub-Criteria	N0.	Factors	ST	LT	ST	LT	
	F511	National Mid/Long-term Development Plan	0.215	0.22	2	2	
	F512	Building Energy Efficiency Promotion and Green Building Development Plan	0.139	0.134	4	4	
Macro Policy	F513	Decisions of Central Government Concerning Building Energy Efficiency Promotion	0.159	0.22	3	2	
,	F514	Industry-Supporting Policies	0.407	0.352	1	1	
	F515	Developing trend of Building Energy Efficiency on Domestic/International Level	0.08	0.075	5	5	
	F521	Financial Subsidy	0.207	0.245	2	2	
	F522	Tax Reduction	0.207	0.137	2	3	
Monetary Policy	F523	Tendency Loan	0.109	0.079	4	5	
	F524	Reduction of Cost of Infrastructure	0.109	0.137	4	3	
	F525	Floor-Area Ratio Reward	0.368	0.402	1	1	
	F531	Industrialization Policy	0.155	0.346	3	1	
	F532	Assessment and Evaluation system concerning governments	0.333	0.25	1	2	
Policy Mechanism Design	F533	Market Access System	0.22	0.165	2	3	
Toncy Weenanishi Design	F534	Technical Encouragement and Reward Policies	0.121	0.091	4	4	
	F535	Top Runner Institution	0.108	0.091	5	4	
	F536	Copyrights Protection	0.063	0.057	6	6	
	F541	Revision of Codes and Standards	0.558	0.588	1	1	
Technology Policy	F542	Revision of Construction Method and Engineering Atlas	0.32	0.323	2	2	
	F543	List of Recommended Products and Technologies	0.122	0.089	3	3	
	F211	Energy-Saving Consciousness of Relevant Enterprises	0.172	0.123	2	3	
	F212	Policy Environment Concerning Energy Saving	0.486	0.465	1	1	
Market Vitality	F213	Availability of Financial Products and Services	0.123	0.214	4	2	
	F214	Market-based Credit System	0.141	0.123	3	3	
	F215	Third-party institutions Development Concerning Energy Services	0.078	0.075	5	5	
	F221	Energy-Saving Needs of Homebuyers	0.205	0.378	2	1	
	F222	Governmental Needs of Conducting Work	0.341	0.229	1	2	
Naada	F223	Enterprises' Needs of Making Profit	0.191	0.151	3	3	
Ineeus	F224	Energy-Saving Needs of Occupants	0.118	0.102	4	4	
	F225	Energy-Saving Needs from Energy Suppliers	0.07	0.051	6	6	
	F226	Energy-Saving Needs from Energy Users	0.075	0.089	5	5	
	F231	Trading Mode	0.092	0.086	4	4	
Marilast Marilaniana	F232	Financing Channel	0.295	0.468	2	1	
Market Mechanism	F233	Business Mode	0.295	0.211	2	3	
	F234	Energy Efficiency Trading Mechanism	0.317	0.234	1	2	
	F241	Stability and Continuity of Policies	0.625	0.683	1	1	
Market Expectation	F242	Market Potential of Renovation	0.238	0.2	2	2	
-	F243	Evaluation of Market Risks	0.136	0.117	3	3	

Table A3. Local weights and rankings of factors.

	No		Local	Ranking		
Sub-Criteria	N0.	Factors	ST	LT	ST	LT
	F311	Technology R&D Cost	0.182	0.121	1	3
	F312	Labor Cost	0.034	0.033	10	9
	F313	Level of Marketization of Technology and Product	0.175	0.203	2	1
	F314	Level of Massive Development of Technology and Product	0.081	0.179	8	2
	F315	Financing Cost	0.094	0.083	4	6
Cost	F316	Promotion of Comprehensive Design Capacity Requirement	0.094	0.1	4	4
	F317	Promotion of Construction Requirement	0.094	0.083	4	6
	F318	Technology Application Cost	0.101	0.064	3	8
	F319	Technology Designing Cost	0.051	0.033	9	9
	F310	Technology Maintenance Cost	0.094	0.1	4	4
	F321	Reduction of Energy Consumption	0.19	0.351	2	2
Ronafit	F322	Length of Payback Period	0.515	0.371	1	1
Denent	F323	Promotion of Building's Value	0.19	0.209	2	3
	F324	Energy-Saving Consciousness of Occupants	0.105	0.07	4	4
	F411	Insulation Performance of Non-Transparent Envelope	0.204	0.139	3	4
	F412	Insulation Performance of Transparent Envelope	0.364	0.301	1	1
Building Material	F413	Airtightness of Windows	0.221	0.262	2	2
C C	F414	New type of wall structure	0.087	0.089	5	5
	F415	Solar Shading Devices	0.124	0.21	4	3
	F421	Appliances of higher energy efficiency	0.375	0.145	1	4
	F422	Choice of Different types of Energy	0.073	0.167	5	3
Building Services and Appliance	F423	Optimized Parameter Settings of Devices	0.215	0.253	2	2
	F424	High Efficiency Component of HVAC system	0.215	0.29	2	1
	F425	High Efficiency Lighting system	0.121	0.145	4	4
	F431	Thickness of Insulation Material	0.053	0.059	7	7
	F432	Types of Windows	0.179	0.23	1	1
	F433	Window–Wall Ratio	0.179	0.166	1	3
Design	F434	Design of Solar Shading System	0.043	0.038	8	8
Deorgin	F435	Limit of Building Shape Coefficient	0.097	0.065	5	6
	F436	Natural Ventilation Design	0.179	0.198	1	2
	F437	Facing Direction of the Building	0.172	0.133	4	4
	F438	Natural Illumination	0.097	0.112	5	5
	F441	Human Behavior Sensors Installations	0.043	0.046	9	8
	F442	Indoor Temperature/Humidity Control	0.041	0.03	10	9
Operation	F443	Indoor Heating/Cooling Energy Consumption Supervision	0.044	0.023	8	10
	F444	Renewable Energy Application on Buildings	0.167	0.266	1	1
	F445	Energy Recycling	0.123	0.176	5	2
	F446	Distributed Energy Utilization	0.074	0.11	7	4
	F447	New Heating/Cooling sources Utilization	0.141	0.072	2	6

Table A3. Cont.

Table A3. Cont.

Sub-Criteria Construction Renovation/Dismantle District Guidance Cooperation Implementation Supervision Assessment	N		Local	Weight	Ranking	
Sub-Criteria	No.	Factors	ST	LT	ST	LT
	F448	Building Energy Consumption Supervision	0.086	0.051	6	7
	F449	Comprehensive Utilization of Resources	0.141	0.11	2	4
	F440	Water Saving and Recycling	0.141	0.117	2	3
	F451	Industrialized Production of Green Building Material	0.25	0.4	2	1
Construction	F452	Construction Process Requirement	0.25	0.2	2	2
	F453	Green Construction Technology	0.5	0.4	1	1
	F461	Construction Waste Reduction	0.4	0.21	1	3
Renovation/Dismantle	F462	Construction Waste Recycling	$\begin{tabular}{ c c c c } \hline Local Weight & Ranki \\ \hline ST & LT & ST \\ \hline 0.086 & 0.051 & 6 \\ 0.141 & 0.11 & 2 \\ 0.141 & 0.117 & 2 \\ \hline 0.141 & 0.117 & 2 \\ \hline 0.25 & 0.4 & 2 \\ 0.25 & 0.2 & 2 \\ 0.5 & 0.4 & 1 \\ \hline 0.4 & 0.21 & 1 \\ 0.4 & 0.55 & 1 \\ 0.2 & 0.24 & 2 \\ \hline 0.044 & 0.043 & 6 \\ 0.182 & 0.163 & 3 \\ 0.281 & 0.239 & 1 \\ 0.066 & 0.061 & 5 \\ 0.281 & 0.37 & 1 \\ 0.145 & 0.123 & 4 \\ \hline 0.2 & 0.392 & 2 \\ 0.2 & 0.392 & 2 \\ 0.2 & 0.392 & 2 \\ 0.2 & 0.278 & 2 \\ 0.4 & 0.165 & 1 \\ 0.2 & 0.386 & 1 \\ \hline 0.5 & 0.5 & 1 \\ 0.5 & 0.5 & 1 \\ 0.5 & 0.5 & 1 \\ 0.5 & 0.5 & 1 \\ 0.493 & 0.4 & 1 \\ 0.311 & 0.4 & 2 \\ 0.196 & 0.2 & 3 \\ \hline 0.2 & 0.25 & 2 \\ 0.4 & 0.5 & 1 \\ 0.4 & 0.25 & 1 \\ \hline 0.5 & 0.571 & 1 \\ 0.25 & 0.286 & 2 \\ 0.25 & 0.143 & 2 \\ \hline 0.26 & 0.163 & 3 \\ 0.327 & 0.297 & 2 \\ 0.413 & 0.54 & 1 \\ \hline \end{tabular}$	1		
	F463	Technology R&D of Construction	0.2	0.24	2	2
	F471	Reclaimed Water Utilization	0.044	0.043	6	6
	F472	Rain Water Collection and Utilization	0.182	0.163	3	3
District	F473	Underground Space Utilization	0.281	0.239	1	2
District	F474	Heat Island Effect Reduction	0.066	0.061	5	5
	F475	District Energy Utilization Plan	0.281	0.37	1	1
	F476	Distributed Energy Utilization	0.145	0.123	4	4
	F111	Guidance of Energy-Saving Conscious	0.2	0.392	2	1
Cuidance	F112	Cultivation of Market Development	0.2	0.278	2	2
Guidance	F113	Setting up Implementation Standard	0.4	0.165	1	3
	F114	Guidance of Value Chain Perfection	0.2	0.165	2	3
Cooperation	F121	Cooperation among market entities adjusted by governments	0.5	0.614	1	1
Cooperation	F122	Cooperation between Governmental Departments	0.5	0.386	1	2
Implementation	F131	Capacity of Planning and Design	0.5	0.5	1	1
mplementation	F132	Capacity of Construction	0.5	0.5	1	1
	F141	Supervision of Construction Quality	0.493	0.4	1	1
Supervision	F142	Credit Supervision of Market Entities Behavior	0.311	0.4	2	1
	F143	Behavior Criterion for Market Entities	0.196	0.2	3	2
	F151	Capacity of Assessing Construction Quality	0.2	0.25	2	2
Assessment	F152	Capacity of Assessing Energy Saving	0.4	0.5	1	1
	F153	Capacity of Energy Efficiency MRV	0.4	0.25	1	2
	F161	Capacity of Facility Management	0.5	0.571	1	1
Operation	F162	Technical Capacity of ESCOs	0.25	0.286	2	2
	F163	Maintenance Capacity of Energy Supply Enterprises	0.25	0.143	2	3
	F171	Innovation of Cooperation Mode Promoted by Governments	0.26	0.163	3	3
Innovation	F172	Innovation Capacity of Business Mode	0.327	0.297	2	2
	F173	Innovation Capacity of Technologies and Products	0.413	0.54	1	1

Notes: ST-Short Term; LT-Long Term.

References

- 1. National Bureau of Statistics of China. *China Statistical Yearbook;* China Statistics Press: Beijing, China, 2018; ISBN 978-7-5037-8587-0. (In Chinese)
- 2. U.S. Energy Information Administration. International Energy Outlook 2014. Available online: https://www.eia.gov/outlooks/archive/ieo14/pdf/0484(2014).pdf (accessed on 10 February 2019).
- Capros, P.; De Vita, A.; Tasios, N.; Papadopoulos, D.; Siskos, P.; Apostolaki, E.; Zampara, M.; Paroussos, L.; Fragiadakis, K.; Kouvaritakis, N. *EU Energy, Transport and GHG Emissions: Trends to 2050, Reference Scenario* 2013; European Commission: Luxembourg, 2013; ISBN 92-79-33728-9.
- 4. Tsinghua University Building Energy Research Center. 2017 Annual Report on China Building Energy Efficiency; China Architecture and Building Press: Beijing, China, 2017; ISBN 978-7-112-20573-8. (In Chinese)
- Huo, T.; Ren, H.; Zhang, X.; Cai, W.; Feng, W.; Zhou, N.; Wang, X. China's energy consumption in the building sector: A Statistical Yearbook-Energy Balance Sheet based splitting method. *J. Clean. Prod.* 2018, 185, 665–679. [CrossRef]
- 6. Development Research Center of the State Concil, World Bank. *Urban China: Toward Efficient, Inclusive, and Sustainable Urbanization;* World Bank Publications: Washington, DC, USA, 2014; pp. 1–583.
- 7. United Nations Environmental Programme, Sustainable Buildings and Climate Initiative. *Buildings and Climate Change Summary for Decision-Makers;* United Nations Environmental Programme, Sustainable Buildings and Climate Initiative: Paris, France, 2009.
- 8. Zhang, M.; Wang, M.; Jin, W.; Xia-Bauer, C. Managing energy efficiency of buildings in China: A survey of energy performance contracting (EPC) in building sector. *Energy Policy* **2018**, *114*, 13–21. [CrossRef]
- Annunziata, E. Energy Efficiency Governance in Buildings: A Multi-Level Perspective; Scuola Superiore Sant'Anna: Pisa, Italy, 2013. Available online: http://www.phdmanagement.sssup.it/documenti/awarded/ Annunziata_Thesis.pdf (accessed on 15 January 2019).
- 10. Ren, S.M.; Guo, H.D.; Xu, Z.-Y. Externality Feature Analysis of Building Energy Efficiency Market and Incentive Policies in China. *Constr. Conserv. Energy* **2009**, *1*, 75–78.
- 11. Zhan, S.L.; Han, Q.M.; Liu, C.B. Analysis of Demanding Factors about Building Energy-Efficiency Market based on Theory of Behavioral Choice. *J. Beijing Jiaotong Univ.* **2009**, *8*, 65.
- 12. He, Z.K.; She, L.Z. Evaluation research on the influencing factors of building energy efficiency promotion based on AHP. *J. Guangzhou Univ.* **2013**, *12*, 76–80. (In Chinese)
- 13. Qu, Y.; Chen, Z.; Darkwa, J. Statistical Analysis of Influence Factors on Building Energy Consumption in China by Literature Overview. *Appl. Mech. Mater.* **2012**, *178–181*, 70–75. [CrossRef]
- 14. Feng, F.; Li, Z.; Ruan, Y.; Xu, P. An Empirical Study of Influencing Factors on Residential Building Energy Consumption in Qingdao City, China. *Energy Procedia* **2016**, *104*, 245–250. [CrossRef]
- 15. Stephan, A.; Stephan, L. Life cycle energy and cost analysis of embodied, operational and user-transport energy reduction measures for residential buildings. *Appl. Energy* **2016**, *161*, 445–464. [CrossRef]
- 16. Liu, Y.; Guo, X.; Hu, F. Cost-benefit analysis on green building energy efficiency technology application: A case in China. *Energy Build.* **2014**, *82*, 37–46. [CrossRef]
- 17. Zhao, X.; Hwang, B.-G.; Tan, L.L.G. Green building projects: Schedule performance, influential factors and solutions. *Eng. Constr. Archit. Manag.* **2015**, *22*, 327–346.
- 18. De Boeck, L.; Verbeke, S.; Audenaert, A.; De Mesmaeker, L. Improving the energy performance of residential buildings: A literature review. *Renew. Sustain. Energy Rev.* **2015**, *52*, 960–975. [CrossRef]
- 19. Chau, C.K.; Leung, T.M.; Ng, W.Y. A review on Life Cycle Assessment, Life Cycle Energy Assessment and Life Cycle Carbon Emissions Assessment on buildings. *Appl. Energy* **2015**, *143*, 395–413. [CrossRef]
- 20. Wen, T.J.; Siong, H.C.; Noor, Z.Z. Assessment of embodied energy and global warming potential of building construction using life cycle analysis approach: Case studies of residential buildings in Iskandar Malaysia. *Energy Build.* **2015**, *93*, 295–302. [CrossRef]
- 21. Mao, G.; Chen, H.; Du, H.; Zuo, J.; Pullen, S.; Wang, Y. Energy consumption, environmental impacts and effective measures of green office buildings: A life cycle approach. *J. Green Build.* **2015**, *10*, 161–177. [CrossRef]
- 22. Doan, D.T.; Ghaffarianhoseini, A.; Naismith, N.; Zhang, T.; Ghaffarianhoseini, A.; Tookey, J. A critical comparison of green building rating systems. *Build. Environ.* **2017**, *123*, 243–260. [CrossRef]

- Nilashi, M.; Zakaria, R.; Ibrahim, O.; Majid, M.Z.A.; Zin, R.M.; Chugtai, M.W.; Abidin, N.I.Z.; Sahamir, S.R.; Yakubu, D.A. A knowledge-based expert system for assessing the performance level of green buildings. *Knowl.-Based Syst.* 2015, *86*, 194–209. [CrossRef]
- 24. Wu, Z.; Shen, L.; Ann, T.; Zhang, X. A comparative analysis of waste management requirements between five green building rating systems for new residential buildings. *J. Clean. Prod.* **2016**, *112*, 895–902. [CrossRef]
- 25. Zuo, J.; Zhao, Z.-Y. Green building research–current status and future agenda: A review. *Renew. Sustain. Energy Rev.* **2014**, *30*, 271–281. [CrossRef]
- Shi, Q.; Yan, Y.; Zuo, J.; Yu, T. Objective conflicts in green buildings projects: A critical analysis. *Build. Environ.* 2016, 96, 107–117. [CrossRef]
- Kong, X.; Lu, S.; Yong, W. A review of building energy efficiency in China during "Eleventh Five-Year Plan" period. *Energy Policy* 2012, 41, 624–635. [CrossRef]
- 28. Li, J.; Shui, B. A comprehensive analysis of building energy efficiency policies in China: Status quo and development perspective. *J. Clean. Prod.* **2015**, *90*, 326–344. [CrossRef]
- Xu, X.; Anadon, L.D.; Lee, H. Increasing Residential Building Energy Efficiency in China: An Evaluation of Policy Instruments. 2016. Available online: https://www.belfercenter.org/sites/default/files/legacy/files/ policy-instruments-residential-building-energy-china-3.pdf (accessed on 5 January 2019).
- U.S. National Renewable Energy Laboratory. Summary of Gaps and Barriers for Implementing Residential Building Energy Efficiency Strategies; U.S. Deparment of Commerce: Springfield, VA, USA, 2010. Available online: https://www.nrel.gov/docs/fy10osti/49162.pdf (accessed on 5 January 2019).
- 31. Zhang, Y.; Wang, Y. Barriers' and policies' analysis of China's building energy efficiency. *Energy Policy* **2013**, 62, 768–773. [CrossRef]
- 32. Shen, L.; He, B.; Jiao, L.; Song, X.; Zhang, X. Research on the development of main policy instruments for improving building energy-efficiency. *J. Clean. Prod.* **2016**, *112*, 1789–1803. [CrossRef]
- Yuan, X.; Zhang, X.; Liang, J.; Wang, Q.; Zuo, J. The Development of Building Energy Conservation in China: A Review and Critical Assessment from the Perspective of Policy and Institutional System. *Sustainability* 2017, 9, 1654. [CrossRef]
- 34. Shen, L.; Zhang, Z.; Long, Z. Significant barriers to green procurement in real estate development. *Resour. Conserv. Recycl.* **2017**, *116*, 160–168. [CrossRef]
- 35. Matisoff, D.C.; Noonan, D.S.; Flowers, M.E. Policy Monitor—Green Buildings: Economics and Policies. *Rev. Environ. Econ. Policy* **2016**, *10*, 329–346. [CrossRef]
- Zou, P.X.W. Risk factor analysis of the Chinese building energy efficiency market using system dynamics methodology. Int. J. Proj. Organ. Manag. 2014, 3, 352–373. [CrossRef]
- Marszal, A.J.; Heiselberg, P. Life cycle cost analysis of a multi-storey residential Net Zero Energy Building in Denmark. *Energy* 2011, *36*, 5600–5609. [CrossRef]
- Deb, K.; Deb, K. Multi-objective Optimization. In Search Methodologies: Introductory Tutorials in Optimization and Decision Support Techniques; Burke, E.K., Kendall, G., Eds.; Springer: Boston, MA, USA, 2014; pp. 403–449. ISBN 978-1-4614-6940-7.
- 39. Roberti, F.; Oberegger, U.F.; Lucchi, E.; Troi, A. Energy retrofit and conservation of a historic building using multi-objective optimization and an analytic hierarchy process. *Energy Build.* **2017**, *138*, 1–10. [CrossRef]
- Saaty, T.L. Fundamentals of the Analytic Hierarchy Process. In *Analytic Hierarchy Process in Natural Resource and Environmental Decision Making*; Schmoldt, D.L., Kangas, J., Mendoza, G.A., Pesonen, M., Eds.; Managing Forest Ecosystems; Springer Netherlands: Dordrecht, The Netherlands, 2001; pp. 15–35. ISBN 978-94-015-9799-9.
- 41. Vaidya, O.S.; Kumar, S. Analytic hierarchy process: An overview of applications. *Eur. J. Oper. Res.* **2006**, *169*, 1–29. [CrossRef]
- 42. Ramanathan, R. A note on the use of the analytic hierarchy process for environmental impact assessment. *J. Environ. Manag.* **2001**, *63*, 27–35. [CrossRef] [PubMed]
- 43. Berardi, U. A cross-country comparison of the building energy consumptions and their trends. *Resour. Conserv. Recycl.* **2017**, *123*, 230–241. [CrossRef]
- Nejat, P.; Jomehzadeh, F.; Taheri, M.M.; Gohari, M.; Abd Majid, M.Z. A global review of energy consumption, CO₂ emissions and policy in the residential sector (with an overview of the top ten CO₂ emitting countries). *Renew. Sustain. Energy Rev.* 2015, 43, 843–862. [CrossRef]

- 45. Adinyira, E.; Kwofie, T.E.; Quarcoo, F. Stakeholder requirements for building energy efficiency in mass housing delivery: The House of Quality approach. *Environ. Dev. Sustain.* **2018**, *20*, 1–17. [CrossRef]
- 46. Mlecnik, E.; Visscher, H.; van Hal, A. Barriers and opportunities for labels for highly energy-efficient houses. *Energy Policy* **2010**, *38*, 4592–4603. [CrossRef]
- 47. Rowe, G.; Frewer, L.J. A Typology of Public Engagement Mechanisms. *Sci. Technol. Hum. Values* **2005**, *30*, 251–290. [CrossRef]
- 48. Weber, E.U. Breaking cognitive barriers to a sustainable future. Nat. Hum. Behav. 2017, 1, 0013. [CrossRef]
- 49. Montibeller, G.; von Winterfeldt, D. Cognitive and Motivational Biases in Decision and Risk Analysis. *Risk Anal.* **2015**, *35*, 1230–1251. [CrossRef]
- 50. Von Winterfeldt, D.; Edwards, W. Decision Analysis and Behavioral Research. 1993. Available online: http://agris.fao.org/agris-search/search.do?request_locale=en&recordID=XF2015037843&sourceQuery=&query=&sortField=&sortOrder=&agrovocString=&advQuery=¢erString=&enableField=&aggregatorField=(accessed on 14 March 2019).
- 51. Howlett, M. Governance modes, policy regimes and operational plans: A multi-level nested model of policy instrument choice and policy design. *Policy Sci.* **2009**, *42*, 73–89. [CrossRef]
- 52. Zhang, J.; Zhou, N.; Hinge, A.; Feng, W.; Zhang, S. Governance strategies to achieve zero-energy buildings in China. *Build. Res. Inf.* **2016**, *44*, 1–15. [CrossRef]
- 53. Treib, O.; Bähr, H.; Falkner, G. Modes of Governance: A Note Towards Conceptual Clarification. *J. Eur. Public Policy* **2005**, *14*, 1–20. [CrossRef]
- 54. Howlett, M.; Ramesh, M. Studying Public Policy: Policy Cycles and Policy Subsystems. *Am. Political Sci. Assoc.* **2009**, *91*, 548–580.
- 55. He, B.; Zhang, X.L.; Jiao, L.D.; Shen, L.Y. China's Ongoing Policy Instrument for Building Energy Efficiency: Drives, Approaches and Prospects. In Proceedings of the 21st International Symposium on Advancement of Construction Management and Real Estate; Springer: Singapore, 2018.
- Saaty, T.L. What is the Analytic Hierarchy Process? In *Mathematical Models for Decision Support*; Mitra, G., Greenberg, H.J., Lootsma, F.A., Rijkaert, M.J., Zimmermann, H.J., Eds.; Springer: Berlin/Heidelberg, Germany, 1988; pp. 109–121.
- 57. Partovi, F.Y.; Burton, J. An analytical hierarchy approach to facility layout. *Comput. Ind. Eng.* **1992**, 22, 447–457. [CrossRef]
- 58. Muralidhar, K.; Santhanam, R.; Wilson, R.L. Using the Analytic Hierarchy Process for Information System Project Selection. *Inf. Manag.* **1990**, *18*, 87–95. [CrossRef]
- 59. Chen, C.F. Applying the analytical hierarchy process (AHP) approach to convention site selection. *J. Travel Res.* **2006**, *45*, 167–174. [CrossRef]
- 60. Honadle, B.W. A Capacity-Building Framework: A Search for Concept and Purpose. *Public Adm. Rev.* **1981**, *41*, 575–580. [CrossRef]
- 61. Greatorex, M. Questionnaire Design; John Wiley & Sons, Ltd.: Hoboken, NJ, USA, 2015.
- 62. Zhou, C.; Yu, L. Legislation Experience of Developed Countries. *Shanghai Real Estate* **2010**, *1*, 50–52. (In Chinese)
- 63. Zhu, D. *Research Evaluation of Building Energy Conservation Policies;* Beijing Communication University: Beijing, China, 2014. (In Chinese)
- 64. Li, S. Role of Government and Its Function in Building Energy Promonion; Qingdao University: Qingdao, China, 2008. (In Chinese)
- 65. Energy World. *Analysis of Green Building Development in* 2013. 2014. Available online: http://www.chinagb. net/news/waynews/20140117/103758.shtml (accessed on 12 February 2019). (In Chinese)
- 66. Zhao, Z.Y.; Chang, R.D.; Zillante, G. Challenges for China's energy conservation and emission reduction. *Energy Policy* **2014**, *74*, 709–713. [CrossRef]
- 67. Yanbing, K.Y. Obstacles and related policy recommendations for promoting building energy efficiency in China. *Heat. Vent. Air Cond.* **2006**, *36*, 33–36.
- 68. Gao, L.; Porter, A.L.; Wang, J.; Fang, S.; Zhang, X.; Ma, T.; Wang, W.; Huang, L. Technology life cycle analysis method based on patent documents. *Technol. Forecast. Soc. Chang.* **2013**, *80*, 398–407. [CrossRef]

- 69. Luo, L.; Zhu, S.; Fu, S.; Zang, Y.; Economics, S.O. Impact of Low Carbon Development on China's Energy Price Reform. *Teach. Res.* **2015**, *5*, 14–23.
- Kagan, R.A.; Axelrad, L. Adversarial legalism: An international perspective. In *Comparative Disadvantages?* Social Regulations and the Global Economy; Brookings Institution Press: Washington, DC, USA, 1997; pp. 146–202.



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).